

Review



State-of-the-Art Literature Review of Power Flow Control Methods for Low-Voltage AC and AC-DC Microgrids

Pavel Ilyushin ^{1,*}, Vladislav Volnyi ², Konstantin Suslov ^{2,3} and Sergey Filippov ¹

- ¹ Department of Research on the Relationship between Energy and the Economy, Energy Research Institute of the Russian Academy of Sciences, 117186 Moscow, Russia
- ² Department of Hydropower and Renewable Energy, National Research University "Moscow Power Engineering Institute", 111250 Moscow, Russia
- ³ Department of Power Supply and Electrical Engineering, Irkutsk National Research Technical University, 664074 Irkutsk, Russia
- * Correspondence: ilyushin.pv@mail.ru

Abstract: The development of AC distribution systems provides for the seamless integration of low-voltage microgrids with distributed energy resources (DERs). This poses new challenges for the control of normal, emergency, and post-emergency states of microgrids, calling for the creation and development of information and communications technology infrastructure. Power converters/inverters that are used to integrate renewable DERs lack inertia. Along with them, fossil fuel-fired generation units are also being integrated into microgrids. These include gas generator sets, diesel generator sets, and microturbines, having small (up to 1-2 s) values of mechanical inertia constants-Tj. This leads to an increase in the rate of transients by a factor of 5–10. Under these conditions, the technical requirements for the speed of automatic power flow control systems, as well as the methods they rely on, have to be reconsidered. Microgrids include DC microgrids, AC microgrids, and hybrid (AC-DC) microgrids. In the case of hybrid microgrids, DERs are connected to the DC grid and are integrated into the AC grid through a common inverter. The complexity of the task of microgrid control is due to the need to choose properly the type and extent of control actions so as to prevent the emergence and development of accidents. The employed control methods must ensure the reliable power supply to consumers and the quality of power in microgrids, as well as the reliable operation of the external distribution systems into which they are integrated. The article gives an overview of control methods for low-voltage AC and AC-DC microgrids, which allow one to tackle effectively solve the tasks.

Keywords: microgrid; distributed energy resources; generating set; renewable energy sources; inverter; converter; power flow control; control action

1. Introduction

Decentralization of generation in the power systems of many countries around the globe is one of the trends of their development, that makes them live up to the environmental, socio-economic, and ethical expectations on the part of society [1]. Sustainable energy development is possible through the creation of modern technologies in the field of distributed energy resources (DERs).

DERs [2] include generation units based on renewable energy sources (RES): photovoltaic modules, wind turbines, etc. DERs also include fossil fuel-fired generation units—gas generator sets (GGS), diesel generator sets (DGS), microturbines, as well as energy storage systems (ESS) of various types, and fuel cells (FC) [3].

Integration of different DERs into distribution systems makes it possible to form small power systems, i.e., microgrids [4]. This allows mitigating the issues resulting from the separate use of DERs, including renewable DERs whose power generation is intermittent in



Citation: Ilyushin, P.; Volnyi, V.; Suslov, K.; Filippov, S. State-of-the-Art Literature Review of Power Flow Control Methods for Low-Voltage AC and AC-DC Microgrids. *Energies* **2023**, *16*, 3153. https://doi.org/ 10.3390/en16073153

Academic Editors: Davide Poli and Davide Fioriti

Received: 11 February 2023 Revised: 24 March 2023 Accepted: 28 March 2023 Published: 31 March 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). nature, ESSs and fuel cells with high capital costs, as well as GGSs, DGSs, and microturbines with high maintenance and repair costs [5].

Reliability of power supply to consumers, power quality, and economic feasibility in microgrids is ensured by the implementation of control algorithms that take into account the features specific to different types of DERs. One of the features is the lack of inertia in renewable DERs as well as ESS and fuel cells [6,7] due to their integration into microgrids through power converters/inverters with almost zero inertia. Moreover, GGSs, DGSs, and microturbines have low values (1–2 s) of mechanical inertia constants—Tj, which leads to an increase in the rate of transients by 5–10 times [8]. Flexibility in microgrids is provided by the implementation of various control algorithms adapted to both grid-connected mode (operating in parallel to the external distribution grid) and islanded mode [9,10]. Switching of microgrids from one mode to another must take place without disturbing the operation of DERs and consumers. This is possible with the bumpless switching implemented by the static switch [11]. This requires a revision of the technical requirements for the speed of load-dispatching devices and the methods they use [12].

Microgrids can be built only on the basis of DC, AC, and AC-DC distribution systems [13]. The variety of types and configurations of networks requires the development of new principles for designing automatic control systems (ACS) and power flow control methods [14]. DC microgrids require a new grid infrastructure to supplement the AC grids in operation, which is costly. The article studies the principles of design of ACSs for AC and AC-DC microgrids. In the latter case, the DERs are combined over a DC grid, but are integrated into the AC grid through a common inverter [15].

Microgrids come in many different network topologies, such as radial, ring-shaped, and mesh structures, each with its advantages and disadvantages [16]. The most widespread is the radial network topology, on the basis of which open-ring circuits are implemented [17]. The mesh topology requires the construction of additional transmission lines, with which multiple network configurations can be implemented in microgrids. However, this contributes to the greater complexity of the ACS of microgrids, as it leads to changes in fault current levels, bi-directional power flows, load re-allocation between DERs, etc. [18].

Available review articles on the subject focused on its different facets, for example, the analysis of engineering solutions and methods of microgrid power flow control with their key advantages and downsides examined. Ref. [19] discussed the control principles of microgrids with different DERs. The study covered their mathematical models and recommendations for application in DC, AC, and AC-DC microgrids. However, the article failed to address the technical limitations of the considered control methods in existing distribution grids. Ref. [20] analyzed the structures of centralized, decentralized, distributed, and hierarchical control systems for microgrids, with examples of implementation of the most common control methods. That being said, the article lacked an analysis of microgrid structures and technical limitations on the application of the considered control methods. Ref. [21] examined the structures of microgrids, taking into account different aspects of their operation, and providing recommended practices for the implementation of various control methods. Ref. [22] presented a classification of the methods of primary control applied to microgrids. The classification was based on the (un)availability of communication links, with the advantages and downsides of each method identified. However, the implementation of the reviewed microgrid control methods was considered only in the case of AC grids. Ref. [23] addressed the issue of implementing a coordinated approach to the control of DERs, including ESSs, in the AC grid. The study also listed the technical requirements to be met by DERs to be able to control their voltage when operating as part of microgrids. In addition, Ref. [23] presented a comparative analysis of various control and optimization algorithms in microgrids, based on which control methods are adapted. However, the scope of the study was limited to considering only centralized and decentralized ACSs of microgrids. Ref. [24] provided a comprehensive review of AC-DC microgrid control schemes, indicating the effect of the control schemes adopted on the stability of microgrids but with the scope limited to AC-DC microgrids only. Thus, the existing review articles

have addressed a large number of important issues, but there remain those that will benefit from additional analysis. In other words, the existing literature reviews considered various algorithms and methods of control of microgrids, which in most cases were presented as having a simplified topology, without taking into account the technical aspects that limit their application.

This article analyzed the state-of-the-art principles of microgrid design that influence the choice of microgrid power flow control methods, as well as power flow control methods themselves. The approach adopted here allows for streamlining the process of designing microgrid ACSs in existing distribution grids' low-voltage AC and AC-DC microgrids. The article aims to discuss the principles of step-by-step design of microgrid control systems based on already-known approaches so as to prevent the improper application of algorithms and methods in the design process.

Section 2 of the article deals with the principles of design and the main types of microgrids. Section 3 describes the principles of design of microgrid control systems, highlighting the main objectives behind their creation and the requirements stipulated by the standards. In addition, we discuss the existing structures for building microgrid control systems and analyze various approaches to their implementation. Section 4 gives an overview of available algorithms and methods for controlling AC microgrids and AC-DC microgrids (hybrid microgrids). Section 5 discusses practical examples of AC microgrid and hybrid microgrid control implementations. Section 6 covers the application of algorithms and methods for controlling AC microgrids, taking into account their optimal use. Section 7 provides recommendations for implementing microgrid control systems and outlines future research in microgrid control. Section 8 wraps up the paper with conclusions and the results of the review performed.

2. Microgrid Design Principles

Integration of microgrids into distribution systems has a positive effect on power systems in terms of improved observability, controllability, power quality, and reliability of power supply to consumers [25,26]. One of the challenges is to ensure the dynamic stability of DERs in emergency and post-emergency states, which is due to the lack of small values of T_j in DERs as this imposes restrictions on allowable load flows and control algorithms of microgrids [27]. Therefore, microgrids must operate as subsystems within the distribution grid control system. Subsystems must have a structure that meets the requirements for ACSs of DERs in order to maximize their technical and economic performance.

2.1. Ways to Integrate DERs

The effective use of different DERs with heterogeneous characteristics is only possible with predefined and standardized procedures for forming microgrids. In what follows we consider the DC, AC, and AC-DC circuits used in the design of microgrids [28,29]. Discussions about the predominant use of one or another circuit have continued until recently [30]. Lessons learned from designing microgrids indicated that the determining factor is the type of current consumed by electrical loads, most of which are powered by an alternating current. It is converted from alternating current to direct current in a number of electrical loads, such as electric cars, lighting systems, ESSs, etc. All DERs can be divided into two groups with respect to the way they are integrated into microgrids. The first group is the DERs that are coupled directly, not requiring the use of power converters. The group includes GGSs, DGSs, and wind turbines of the first, second, and third types [31]. The second group includes DERs integrated through inverters (converters). This group includes PV modules, Type 4 wind turbines [31,32], microturbines, ESSs, and fuel cells. Therefore, AC operation is preferable for the former group and DC operation for the latter. This eliminates the power converter from the circuit, which helps reduce power losses and capital costs, and increases the reliability of DERs.

2.2. Optimal Structure of DERs in Microgrids

Key defining features of microgrids include the large-scale integration into them of renewable DERs, as well as heterogeneous loads with intermittent power generation/consumption. This can cause short-term unbalances of active and reactive power, which leads to decreased power quality parameters and unstable operation of microgrids [33,34]. To solve this problem, it is effective to use ESSs, but the capital and operating costs of ESSs remain high. It is required to determine the allowable power of renewable DERs as part of microgrids, as well as the optimal power and energy capacity of ESSs [35].

A proper design is paramount to the reliability of microgrids because when they operate in the islanded mode it allows them to maintain the specified power quality parameters with respect to voltage and frequency. To model uncertainties when scheduling microgrid power flows, it is proposed to use a probabilistic power flow, usually calculated by Monte Carlo simulation, which is computationally expensive [36]. An alternative to the above method is the use of approximated and improved iterative algorithms [37], which have good accuracy in estimating variables and probabilistic parameters while requiring much fewer computational resources.

Selecting an optimal structure of DERs in microgrids should be backed by statistical data on load profiles on different days (weekdays, weekends, holidays), values of economic performance metrics of DERs, geographical conditions of DERs location, values of reliability metrics of DERs, etc., as well as requirements for power supply reliability and power quality [38,39].

The use of optimization methods with life cycle cost minimization serving as the criterion makes it possible to determine the allowable capacity of renewable DERs, ESS specifications, as well as the microgrid design scheme [40]. Various methods can be used in the optimization process, the most common are methods based on heuristic algorithms, mathematical programming algorithms, and interior points methods [38]. As a rule, it is possible to achieve the stated goals by using equipment with extensive power flow control capabilities as part of microgrids [40–42].

2.3. Switching Equipment of Microgrids

Microgrids can be integrated into medium- and low-voltage distribution grids, but they are most often connected to low-voltage grids, given the widespread use of PV modules in households. Microgrids are connected to a point of common coupling (PCC) by means of a static switch. The static switch position determines the mode of microgrid operation (grid-connected or islanded) and implements the transition between them [43,44]. The static switch is a fast-acting switching device based on power semiconductor components, such as triacs, which have high overload capacity [45].

Smart circuit breakers are used to provide high-speed control of DERs and critical loads. To control non-critical loads, it is standard circuit breakers with external control that are used more often [46].

2.4. Microgrid Structures

The optimal choice of the microgrid structure reduces power losses and grid modernization costs, ensures efficient frequency and voltage control in the islanded mode, and improves the reliability of power supply to consumers [30]. There are three main structures of microgrids: DC, AC, and AC-DC microgrids [47].

2.4.1. DC Microgrids

Coupling of DERs into microgrids without the use of power converters eliminates power losses that occur in their conversion in the amount of 10–25% [48]. Another advantage of DC microgrids (Figure 1) is that they maintain the values of power quality parameters within the required range, regardless of the AC grid mode of operation [28,49].



Figure 1. DC microgrid structure.

DC microgrids are integrated into the distribution grid through a bi-directional interlinking converter (IC). It provides bi-directional power transmission and galvanic isolation of DC and AC grids. DERs are connected through AC-DC or DC-DC power electronic converters (PEC) to maintain the DC voltage within the required range.

The transition of DC microgrids into the islanded mode takes place in the event of disturbances in the AC grid when the DC grid voltage deviates by more than $\pm 10\%$, and there are ripples of more than $\pm 5\%$ [50].

The advantages of DC microgrids include a simple structure, lower construction costs, independence from the AC grid power flow, ability to maintain the required values of power quality parameters during AC grid accidents, easy voltage regulation, no reactive power, lower power losses, and no need for a synchronization algorithm in the ACS of microgrids.

The downsides of DC microgrids are the need to build a DC grid, the difficulty of integration into the existing AC grid, and the implementation of protection functions due to the lack of standards and hands-on experience [51,52]. Moreover, DC microgrids require the use of special-purpose DC circuit breakers for high fault currents, electric motors are impossible to integrate without the use of AC-DC PECs, reliability of power supply to consumers is compromised due to the presence of a common IC.

2.4.2. AC Microgrids

The availability of existing AC grids makes it much easier to create microgrids based on them. The microgrid structure proposed in the CERTS Microgrid Concept [53] is commonly used. It provides the ability to switch microgrids to the islanded mode while maintaining the power supply to the load. In this case, the main circuit breaker is connected to the substation low voltage bus, and all DERs are connected to the microgrid via DC-AC and AC/DC-DC/AC PECs [54,55] (Figure 2).

Figure 2 shows that there are three feeders in the microgrid: feeder 1 and feeder 2 serve the critical load, and feeder 3 serves the non-critical load. The installed circuit breakers allow the microgrid to be reconfigured to balance active and reactive power. The static switch controls the mode of operation of the microgrid with the distribution grid, providing switching to/from the islanded mode. Criteria for the switching of microgrids into the islanded mode are frequency deviations of ± 0.2 Hz and voltage deviations of $\pm 5\%$, as well as overcurrent of the transmission line equal to 30% [56,57]. In the islanded mode of operation, critical loads are fed from DERs, and non-critical loads are fed from the external distribution system.



Figure 2. AC microgrid structure.

The operation of microgrids in the grid-connected mode contributes to the reliability of power supply to consumers and facilitates load dispatching of DERs during load surges/shedding [58]. The disadvantages include the presence of a large number of PECs, which compromises the reliability and efficiency of AC microgrids. There are engineering solutions that are known to improve the reliability of PECs, but their adoption is not ubiquitous [59].

Considering the above, the AC microgrid structure is most in demand for integrating DERs into existing distribution grids.

2.4.3. Hybrid (AC/DC) Microgrids

The joint use of DC and AC microgrids allows for a better structure of a hybrid AC-DC microgrid to be implemented. This simplifies the integration of different DERs and DC and AC consumers and contributes to the efficiency of their operation [28]. This case is shown in Figure 3.



Figure 3. Structure of the AC-DC microgrid.

In a hybrid microgrid, AC consumers are connected directly to the AC bus and DC consumers through DC-DC PECs to adapt the voltage level at the bus to the voltage level of the load. Connecting DERs and DC loads to the DC bus minimizes the effect of harmonic content on the power quality parameters of the external distribution grid. Hybrid microgrids reduce the total number of PECs, which helps reduce power losses and improve the reliability of microgrids.

Creating a hybrid microgrid requires considerable expenses to build a DC grid and install a static switch to connect the microgrid to the external distribution grid. It breaks even faster as the number of DC DERs that are connected grows, because of the decreasing number of PECs. At the same time, the task of controlling a hybrid microgrid becomes more complicated due to the need to coordinate AC and DC grid control algorithms.

Hybrid microgrids combine the main advantages of DC and AC microgrids, allowing for the optimal use of different DER technologies [60]. The hybrid microgrid structure shown in Figure 3 is one of the possible topologies. Other structures differ in the way AC-DC microgrids are connected, using one or two static switches connected in parallel [55,61,62]. In addition, they can use power routers, solid-state transformers, and small-sized flexible AC transmission systems. However, in hybrid microgrids it is recommended to use a semi-regulated high-frequency resonant DC transformer, operating at a resonant frequency, which provides the required active power transmission ratio with a constant voltage conversion gain [63].

3. Design Principles of Microgrid ACS

One of the main tasks of creating ACSs of microgrids is to ensure optimal control of electricity production and consumption, as well as the sale of electricity as governed by market mechanisms. This enhances the investment appeal of microgrids acting as aggregators of DERs as well as adjacent microgrids [64].

3.1. Tasks of Microgrid Control

The grid-connected and islanded modes of microgrids both have their unique features in terms of control arrangement. In the former mode, the frequency and voltage are determined by the external distribution grid, and in the latter mode, by the frequencyoperated generation unit. In addition, it is necessary to ensure bumpless transfer from one mode to the other and back [65,66].

In the grid-connected mode, the tasks of maintaining the voltage at nodes, overload management, reduction of power losses, as well as maintaining the required values of power quality parameters (reduction of voltage sags, harmonic compensation, flicker mitigation, etc.) are solved. Furthermore, the control of power flows between the microgrid and the external distribution grid by controlling the load of DERs is implemented, as well as a bumpless transfer from the grid connected to the islanded mode by the action of automatic multiparameter partitioning devices [67].

Operation in the islanded mode solves the following problems:

- frequency and voltage regulation within a specified range for all topologies and operating conditions by controlling the active and reactive power of DERs, while respecting the available constraints;
- automatic synchronization with the external distribution grid when the load flow parameters are normalized there;
- electricity sales based on market mechanisms through optimized load dispatching of DERs;
- reliable power supply to critical loads as part of the microgrid for all topologies and operating conditions;
- automatic "black start" of microgrids in case of blackout;
- optimization of operating costs for the generation and distribution of electric power to consumers [68,69].

The above tasks are divided into groups. These groups of tasks [70] are addressed at three levels: the local level, the microgrid level, and the level of an external distribution grid (Figure 4).



Figure 4. Separation by the level of tasks completed in the operation of microgrids.

The following features should be taken into account when completing control tasks:

- power flows: bi-directional power flows occur between microgrids and the external distribution grid, depending on the modes of generation of DERs and consumption within microgrids;
- stability: short-term fluctuations in power flow parameters can occur due to the interaction of different ACSs, as well as when microgrids switch from the grid-connected mode to islanded mode;
- low inertia: the dynamic characteristics of DERs, especially of those connected through PECs, differ significantly from the characteristics of high-power generation units coupled directly. Low inertia in microgrids and lack of the spinning reserve can lead to significant frequency and voltage deviations in the islanded mode;
- uncertainty: it results from the intermittent demand and generation of electricity by renewable DERs, which requires the ACS of microgrids to factor in the current value of generation, predicted electricity demand, and its price to ensure reliability and cost-effectiveness.

Since microgrids have the features that make them stand apart, their design principles should be adapted to the specified operating conditions, which differ from conventional ones in distribution grids (load balance across phases; load constancy; X/R ratios of transmission lines) [71,72].

3.2. Requirements for ACSs of Microgrids

A large number of topologies and operating conditions of microgrids require the design of more advanced ACSs [73]. The creation of service-oriented ACSs is possible through the use of modern information and communications technology infrastructure, intelligent electronic devices, and integrated hardware and software systems [74].

Table 1 [75] lists the regulatory documents containing technical requirements and recommended practices. These documents should be used for the design and operation of microgrids as well as their ACSs, depending on the characteristics of microgrids (capacity; topology; types of DERs; load mix).

Application Scope	Standard	Short Description	Reference
Electrical Safety	IEC 60364-1 «Low-voltage electrical installations—Part 1: Fundamental principles, assessment of general characteristics, definitions»	Recommendations for design and verification of electrical installations of nominal voltages up to 1000 VAC or 1500 VDC to guarantee the safety	[76]
Electromagnetic compatibility	IEC 61000-4-30 Electromagnetic compatibility (EMC)—Part 4–30: Testing and measurement techniques—Power quality measurement methods»	Requirements for power quality boundaries for AC and DC buses (e.g., voltage unbalance is limited to 3%)	[77]
Design of systems with DERs	IEC 61508 «Functional safety of elec- trical/electronic/programmable electronic safety-related systems»	Features of the design of electrical, electronic, and programmable systems that provide the required reliability, efficiency, and fault-free operation of microgrids with DERs	[78]
Structures for control system design	IEC 61499 «Function blocks»	Distributed control structure; standardized requirements for software tools to ensure software compatibility in intelligent devices, machines, and systems	[79–81]
Creation of a general information model	IEC 61970 «Energy management system application program interface (EMS-API)»	Requirements for the general information model, equipment, and other components of the power system in the form of classes, their properties, and links	[82]
	IEC 61968 «Application integration at electric utilities—System interfaces for distribution management»	Requirements for exchanging asset management, work scheduling, and billing data for consumers in microgrids	[83]
Connecting microgrids to the public grid	IEEE 1547 «Standard for Interconnection and Interoperability of Distributed Energy Resources with Associated Electric Power Systems Interfaces»	Rules for the safe integration of DERs (up to 10 MVA), allowable load flows during microgrid switching to the islanded mode, and requirements for power quality boundaries. Standardized values of <i>U</i> , <i>f</i> , and phase angle at the interface point	[56]
Power utility automation	IEC 61850 «Communication networks and systems in substations»	Rules for the communication between microgrids and substations, as well as intelligent devices within microgrids	[84]
Information security	IEC 62351 «Power systems management and associated information exchange—Data and communications security»	Requirements for information security (data transfer and communications design)	[85]
Power System Management (PSM)	IEC TR 62357 «Power systems management and associated information exchange»	Requirements for power system management processes and related information exchange	[86,87]

 Table 1. Standards for the design of microgrids.

Application Scope	Standard	Short Description	Reference
Energy Management System (EMS)	IEEE Std P2030 «Guide for Smart Grid Interoperability of Energy Technology and Information Technology Operation with the Electric Power System (EPS), End-Use Applications, and Loads»	EMS-based control layer functions that are common to all microgrids, regardless of their structure, topology, or affiliation	[88]
Stand-alone DC power suppliers	IEEE 2030.10 «Standard for DC Microgrids for Rural and Remote Electricity Access Applications»	The rules of operation of 48 VDC microgrids in self-sustaining communities. Recommendations on DC grid control and communication protocols	[89]
Medium voltage DC bus	IEEE 1709 «Recommended Practice for 1 kV to 35 kV Medium-Voltage DC Power Systems on Ships»	Recommendations for maintaining power quality parameters in 1000 V to 35,000 V grids within specified boundaries (e.g., maximum acceptable ripple and DC voltage tolerances)	[50]
Energy thermal efficiency of buildings	ISO 52016-1 «Energy performance of buildings—Energy needs for heating and cooling, internal temperatures and sensible and latent heat loads—Part 1: Calculation procedures»	Response time requirements in low-voltage AC microgrids to meet thermal performance requirements for buildings (e.g., energy requirements for heating and cooling)	[90]
Connection of electric vehicles	IEC 61851 «Electric vehicle conductive charging system—Part 23: DC electric vehicle charging station»	Information about household electric vehicle charging stations in single-phase (250 V) and three-phase (480 V) systems	[91]

Table 1. Cont.

3.3. Structures of Microgrid ACSs

It is impossible to create a standardized design of a microgrid ACS [70]. This is due to the wide variety of DERs, microgrid structures, types of electrical loads, etc. Therefore, the structure of an ACS is formed on a case-by-case basis for each microgrid during its design.

The same approaches are used in the design of microgrid ACSs as those in the ACSs of power systems. The distribution management system (DMS) is commonly used in distribution grids. The DMS implements algorithms for monitoring the grid and DERs, as well as those of grid reconfiguration and voltage regulation by controlling static reactive power compensators and the largest generation units. However, the DMS does not provide for the implementation of control actions on multiple DERs, which leads to compromised reliability of distribution grids. When DERs are deeply integrated, controllability can only be achieved by creating more advanced ACSs of microgrids [92].

The microgrid structure usually has a central controller as well as controllers of DERs. Most often local controllers that combine their functions are used instead of the latter [93–96].

Local controllers come either as separate devices or as a set of functions in intelligent electronic devices (smart meters, PECs) with sufficient computing power. Local controllers implement algorithms for monitoring and control, including the control of the power of DERs, through control actions on PECs.

Information and communications technology infrastructure are not necessary for local controllers to function. They implement decentralized algorithms based on frequency and voltage droop control. This allows for the DERs power control based on local data, since

the frequency is related to active power and voltage is related to reactive power [93]. This approach is applicable to DERs based on synchronous generators, but its implementation in DERs with PECs is not possible, given the features discussed below.

The central controller coordinates local controllers in the microgrid to improve technical and economic performance by implementing ancillary services and ensures microgrid communication with the energy service company and distribution system operator. In addition, the central controller calculates and issues setpoints and control actions to local controllers. The central controller is usually designed as a separate unit installed in the medium voltage substation but is sometimes integrated into the DMS [70].

3.3.1. Hierarchical Control in Microgrids

A large number of different DERs with different protection setpoints and allowable operating ranges influence the nature and parameters of transients, which determines the requirements for control algorithms in ACS microgrids [97]. It is important to note that the different control algorithms are decoupled in time, since they require to be executed in a certain sequence. Therefore, in microgrids, ACSs with a hierarchical structure, which is also standardized, are widely used [98].

Hierarchical ACSs can have a different number of control levels. The most common are the three-level ACSs:

- primary level: power, voltage, and current monitoring of DER. The level implements the basic algorithms with setpoints specified by higher-level controllers via PECs [99];
- secondary level: monitoring of the execution of algorithms on the first level and maintaining the required values of power quality parameters in the microgrid [100]. The level implements algorithms for microgrid switching from the grid-connected mode to islanded mode and back, as well as control of the amount of power flows to and from the external distribution grid (other microgrids);
- tertiary level: implements optimization algorithms to improve the economic performance of microgrid operation. This requires information and communications technology infrastructure, as well as intelligent algorithms for decision-making [101].

Thus, the tasks of power supply reliability and power quality are implemented at the primary and secondary levels, and economic performance is ensured at the tertiary level of control.

Separating control tasks by the time of their execution streamlines the analysis of microgrid behavior, as well as modeling and forecasting processes. The transition to higher-level tasks is accompanied by a decrease in the speed of algorithm execution [102] (Figures 5 and 6).



Figure 5. Timeline of the execution of main control algorithms in microgrids.



Figure 6. Hierarchical control of AC-DC microgrids in a building.

Coordinated management of DERs in microgrids can be implemented using centralized, decentralized, or distributed approaches to control (Figure 7). The choice of a particular approach is determined by the goals of the creation of the microgrid and the specifics of microgrid design, as well as the availability or accessibility of resources (personnel, equipment, etc.).



Figure 7. Approaches to microgrid control.

The degree of decentralization of control in microgrids depends on the functionality, the control algorithms implemented, as well as the computing power of local controllers.

3.3.2. Centralized Control of Microgrids

The centralized approach to the arrangement of microgrid control (Figure 7a) is wellresearched because its implementation is based on the same principles as in the hierarchical ACS. In the case of centralized control, all control functions are implemented in the central microgrid controller, which analyzes the information from local controllers of DERs and calculates and issues the control actions. The control actions are implemented by local controllers, whose functions in AC and AC-DC microgrids are often performed by PECs.

Achieving maximum efficiency in microgrids is possible through coordinated control of all DERs, which is attainable when centralized control is implemented. More DERs operating in a microgrid require more processing power from the central controller, so this approach to control is used in small microgrids.

Having one device performing all calculations in a microgrid requires reliable highspeed broadband communication links between local controllers and the central controller. However, this does not guarantee high reliability of the entire ACS, since the employed topology of the communication network (point-to-multipoint) in case of failure of the communication interface of the central controller, leads to a complete failure of the ACS.

3.3.3. Decentralized Control of Microgrids

In the decentralized approach (Figure 7b), all control functions are implemented in local controllers using only local information. There is no need for a central microgrid controller and information and communications technology infrastructure. There is implicit communication between the local controllers. It is due to the mutual influence of the results of the implementation of algorithms by individual local controllers on the change of load flow parameters in the microgrid. This reduces the computational load on the local controllers since the completion of the control task is divided among the individual components.

3.3.4. Distributed Control of Microgrids

The presence of two-way communication between neighboring local controllers allows for the creation of the distributed ACS of microgrids (Figure 7c) to maximize the effect of optimization algorithms.

The partial mesh topology of the communication network used in distributed control allows it to remain operational in case of failure at any node. This communication network topology simplifies the process of scaling it up since each individual node is essentially a router. When connecting new DERs, it is not necessary to reconfigure the communication network, and this allows for the use of plug-and-play technology. The distributed control approach is more attractive than the centralized approach, presenting a holarchic control architecture [103] with a holonic production system [104].

The hybrid approach to microgrid ACS implementation is to combine local controllers of one cluster (several DERs connected close to each other) under the control of one central controller connected to the central controller of another cluster in the microgrid. This improves the coordination and efficiency of DER control in both grid-connected and islanded modes [105].

4. Algorithms and Control Methods for Microgrids

4.1. Control Methods for PECs

Ref. [30] provided recommendations for the choice of structures and methods of microgrid control using different communication links. The choice should take into account the structure of the grid as well as the allowable power flows of microgrids. Existing studies addressed the issue implementation techniques of bumpless switching to the islanded mode, as well as those of economic scheduling, joint regulation of frequency and voltage, improving the reliability of redundancy, preventive and emergency control, synchronization and execution of automatic "black start" have been studied. Ref. [106] reviewed practical examples of the construction and operation of microgrids based on a hierarchical ACS.

Creating microgrids requires solving optimization problems, which fall into three classes: optimal power flow, scheduling, and planning. Various metaheuristic methods have been most widely used for their solution, since they show good results for solving complex control problems, especially those characteristics of microgrids with intermittent generation [107].

Metaheuristic methods include:

- evolutionary computations: methods that simulate the evolution of population members (genetic algorithms, differential evolution);
- methods of swarm intelligence: methods capturing the properties of self-organizing groups of biological organisms with "smart" global behavior (ant colonies, harmony search algorithm, particle swarm optimization, etc.);

- artificial immune systems: methods inspired by theoretical immunology and modeling the processes used by the immune system to respond to external threats;
- non-population-based metaheuristics: methods based on finding a single solution, i.e., temporarily taking the worst solution with a probability that decreases as more iterations are run (simulated annealing, tabu search) [108].

The choice of the metaheuristic method is based on the available computational resources and the number of function convergence estimates [109]. For example, evolutionary computations based on genetic algorithms are applicable to any configurations of microgrids, allow the use of hybrid approaches, are easily scalable, and do not impose restrictions on the functions they perform. However, the performance of the algorithm is determined by the quality of the coding of the optimization problem, as well as by its sensitivity to parameter setting [110]. The differential evolution method, which has a higher rate of function convergence than the genetic algorithm, is a simple and reliable method applied to optimization problems with constraints that require a relatively small number of control variables. However, this algorithm strongly depends on parameter setting, which determines the convergence rate [111].

Issues with coding the algorithm are resolved due to the use of swarm intelligence methods, such as the particle swarm optimization method. This method does not require special coding, and its simplicity combined with efficiency makes it an ideal solution when computational resources are limited [111].

Artificial immune systems are formed on the basis of a distributed model with no control center, i.e., they use only local information. Therefore, these methods require a minimal amount of computational resources, unlike population-based methods. However, these methods require calibration to solve the optimization problem, unlike evolutionary computation and swarm intelligence methods [112].

Non-population-based metaheuristic methods require the least computational resources relative to other methods, but they also have the lowest accuracy [113].

The studies and examples reviewed here show that the key challenge is to ensure the proper operation of the static switch through which the microgrid communicates with the external distribution grid. In the grid-connected mode of operation, it must function as a current source, providing power output and consumption from the external distribution grid, and in the islanded mode, as a voltage source, creating a reference voltage of nominal frequency for the microgrid [114].

4.1.1. Switching Microgrids to the Islanded Mode

The main difficulties in the implementation of the ACS of the static switch are time delays that occur both in the dynamic response of the ACS and during its operation. Compensators of voltage, active and reactive power are used to stabilize the external characteristics of the static switch [115]. For example, when the microgrid switches to the islanded mode, deviations are possible between the actual and reference voltage values of the DC link static switch, in which case the output voltage compensator that sets the reference current for the internal control loop is triggered [14].

4.1.2. Operation of Microgrids in the Islanded Mode

Switching of microgrids to the island mode takes place either in case of accidents or during scheduled switching operations in the external distribution grid. Since the main circuit breaker of the microgrid has high speed, the transfer to the islanded mode requires rapid coordination of control algorithms of all DERs with PECs in the microgrid, so the ACS needs to implement an algorithm to control microgrid operation modes.

The main goals of microgrid islanded mode control are as follows [116]:

- maintaining the necessary voltage value at the voltage source in the microgrid to ensure proportional power allocation between DERs with PECs;
- maintaining the specified frequency at the voltage source in the microgrid, according to the specified characteristic and droop coefficient.

4.1.3. Synchronization of Microgrids with the External Distribution Grid

The main method used to synchronize microgrids with an external distribution grid is the phase-locked loop technique [117]. In this technique, the phase angle of the voltage vector is selected based on the microgrid voltage vector and the q-axis voltage vector at the point of common coupling. Then, using a phase-locked loop, the angle between these vectors at the voltage source is reduced to zero.

However, the phase-locked loop technique cannot be used in the presence of an unbalanced (non-linear) load, as it is a source of interference to the execution of the algorithm. Its operation provokes sudden changes (jumps) in the phase angle. In this case, a modified phase-locked loop method should be used, which additionally uses a second-order generalized integrator to adjust frequency [118]. To coordinate local microgrid controllers relying on voltage and current control algorithms, as well as the phase-locked loop technique, it is necessary to provide for the implementation of control actions from the central microgrid controller or apply other methods [119].

4.2. Features Unique to Microgrid ACS Implementation

4.2.1. AC Microgrids

The microgrid ACS usually has a three-layer hierarchical structure. Primary control in microgrids is implemented by local controllers acting on PECs of DERs, providing control of their active and reactive power and allocation of power among DERs.

Control of DERs is implemented by algorithms that fall into several categories, depending on the selected current and voltage reference system: synchronously rotating system dq, complex system $\alpha\beta$, and natural three-phase system *abc* [120]. In a synchronously rotating system, proportional-integral (PI) regulators are used, in a complex system, proportionalresonance (PR) regulators are used, and in a natural three-phase system, PI or PR regulators are used. Output parameters of DERs are controlled through an external loop (for voltage control) and an internal loop (for current control). Traditionally, linear and nonlinear current regulators are used for this purpose [121].

Linear current controllers are used in PI controllers in DC grids because they can track specified values with zero steady-state error. On the other hand, PR controllers are most often used for control in AC grids.

Non-linear current regulators are based on hysteresis current control, sliding load flow control, neural network algorithms, and fuzzy logic algorithms. Thus, nonlinear regulators control the active and reactive power of DERs by regulating the current in the DC link of PECs. Various problems may arise, which were discussed in detail in [122].

The most common method for controlling PECs of DERs are multi-parameter methods and their various modifications, which improve the dynamic response when the microgrid load flow changes and increase the tolerance to load uncertainty (non-linearity) [123–126].

Managing the allocation of power between DERs helps ensure the reliability of power supply to consumers and the optimal operation of microgrids. To this end, droop-based methods as well as methods that do not use droop are employed.

The droop is specified by the *P*-*f* and *Q*-*f* characteristics in order to regulate the frequency and voltage in the microgrid islanded mode [127–131]. If $U_{mg} < 0$ and $E < \delta$ when integrating DERs through PECs with a small value of the angle δ , the magnitude of active and reactive power when connected to a transmission line with predominantly inductive reactance will be determined by Equations (1) and (2):

$$P = \frac{U \cdot E}{X} \cdot \delta; \tag{1}$$

$$Q = \frac{U \cdot E - U^2}{X},\tag{2}$$

where *U* is the PEC output voltage; *E* is the voltage at the point of common coupling.

Given that in low- and medium-voltage power lines the ohmic resistance prevails, expressions (1) and (2) are not applicable for these grids. The *P-f* droop characteristic reduces the output voltage frequency with increasing output power of PECs of DERs, while the Q-f droop characteristic reduces the output voltage amplitude with increasing reactive power in the microgrid islanded mode.

Specifying the droop allows one to simulate the inertia of the synchronous machine for DERs with PECs in order to maintain the balance of active power and frequency in the islanded mode. In case of emergency disturbances, significant deviations of frequency and voltage from the nominal values may occur, which leads to instability of regulation. For voltage sources, the frequency and voltage values are set according to Equations (3) and (4):

$$\omega = \omega_0 - D_1 \cdot P; \tag{3}$$

$$U = U_0 - D_2 \cdot Q, \tag{4}$$

where ω_0 —nominal angular frequency at the voltage source; U_0 —nominal output voltage of the voltage source; D_1 —droop constant for the *P*-*f* characteristic; D_2 —droop constant for *Q*-*f* characteristic. They can be specified: $D_1 = \frac{\omega_{max} - \omega_{min}}{P_{max}}$ and $D_2 = \frac{U_{max} - U_{min}}{Q_{max}}$, where ω_{max} and ω_{min} are the maximum and minimum angular frequency values; U_{max} and U_{min} are the maximum and minimum voltage values; P_{max} and Q_{max} are the maximum allowable values of active and reactive power of DERs.

The disadvantages of traditional droop-controlled methods were discussed in detail in [130]. Some of the difficulties can be solved by modifying the techniques. Next, we consider modified techniques for setting droop [131] relying on the following:

- static characteristic of voltage regulation as a function of active power (*U-P_s*) and inverse static characteristic of frequency regulation as a function of reactive power (*f-Q_{is}*) [132,133]; static characteristic of reactive power regulation as a function of voltage increment *Q-U'* [134]; static angle regulation [135];
- transformation of the reference frame [136];
- virtual impedance [137,138], virtual inertia [139,140];
- unbalanced control of active and reactive power flows and non-linear load dispatching [141–143];
- adaptive (modified) droop control [144,145].

In what follows, we consider the principles of their implementation, as well as the advantages and disadvantages.

Static characteristics of voltage regulation as a function of active power (U- R_s) and inverse static characteristics of frequency regulation as a function of reactive power (f- Q_{is}) are used in low-voltage grids. In this case, the output voltage of PECs of DERs decreases as the output power increases, while the frequency increases as the output reactive power increases. This improves the power allocation in low-voltage grids but it depends on the parameters of the grid. This approach is not applicable to grids with non-linear loads.

The static characteristic of reactive power regulation as a function of the voltage increment Q-U' provides reactive power allocation between DERs regardless of the transmission line impedance. The voltage restoration circuit maintains a constant output voltage by maintaining a zero-voltage increment U' = 0. This method depends on initial conditions and also has low stability.

Static angle regulation has a significant low-frequency deviation compared to static frequency regulation. The angular phase of the PECs output voltage can be used effectively, provided that low bandwidth communication links are in place. However, the method requires inductive reactance between the PECs and the microgrid AC bus. A larger angle enhances the droop effect through an additional control loop in low-power microgrids in order to increase the accuracy of load allocation between DERs.

The reference system transform method uses a linear orthogonal matrix to convert the values of active and reactive power flows to a new reference system, where they do The virtual impedance is used in the feedback circuit of the voltage regulation loop. Thus, the PECs output voltage is generated by adjusting the virtual impedance $Z_u(s)$. The output voltage of PECs (*U*) is defined as $U = U_0^* - Z_u(s) \cdot i_0$, where U_0^* —output voltage of PECs at no-load; i_0 —output current; Z_u —virtual impedance. If $Z_u(s) = sL_u(s)$, the output voltage decreases in proportion to the derivative of the output current. However, the harmonic distortion of the output voltage will have large values when feeding a non-linear load. To solve this problem, one has to use a high-pass filter instead of sL_u .

The emulation of virtual inertia in microgrids prevents nuisance tripping of circuit breakers during short-term frequency decreases/increases. The droop coefficient is modified as the function $\frac{df}{dt}$, which is effective when $\frac{df}{dt}$ exceeds the threshold value *B*. The rule of the modified droop control method takes the form:

$$\omega_i^* = \omega_n - c_i \cdot (P_{n,i} - P_i), \tag{5}$$

where ω_i and P_i —voltage frequency and output active power of DERs, respectively; indices "*" and "n"—reference and a nominal number of PECs; c_i —droop coefficient, determined by Equation (6):

$$c_{i} = \begin{cases} c_{n,i} - \omega_{1} \cdot \left| \frac{df}{dt} \right|^{\omega_{2}}, \text{ for } \left| \frac{df}{dt} \right| \ge B \\ c_{n,i}, \text{ for } \left| \frac{df}{dt} \right| < B \end{cases},$$
(6)

where $c_{n,i}$ —nominal droop gain, which changes if the rate of frequency change exceeds the set value; ω_1 and ω_2 —frequencies corresponding to the maximum PECs power and the maximum allowable frequency change, respectively.

The virtual inertia emulation allows us to represent the voltage source as an equivalent virtual synchronous machine (VSM) with a static frequency control characteristic in microgrids [140] (Figure 8).



Figure 8. Emulation of inertia by a virtual synchronous machine.

Low-pass filtering of active power in the voltage source circuit also contributes to virtual inertia, which improves stability and damping in microgrids. Simplified frequency droop control *P*-*f* is implemented as per Equation (7):

$$T_f \cdot \frac{1}{m_p} s \cdot \omega^* = P_0 - P_f - \frac{1}{m_p} (\omega^* - \omega_g);$$
⁽⁷⁾

where $P_0 = p_0$ —reference value of active power; T_f —time constant of the low-pass filter; $P_f = p_{el}$ —filtered output active power; $m_p = 1/k_d$, m_p —frequency droop coefficient; k_d — damping coefficient of active power regulator; ω^* —reference value of angular frequency;

 ω_g —angular frequency at the point of connection to microgrid; p_m —active power signal in the feedback link; θ —angular position of VSM rotor.

The negative effect of unbalanced flows of active and reactive power in the transmission line can be compensated by applying a high-frequency signal, which was discussed in detail in [141].

There is a method that provides current-sharing between the PECs of DERs connected in parallel without coupling between them [142]. There is also a technique that has harmonic voltages calculated based on the current harmonics and impedance of the PECs for each frequency. The resulting voltage is then added to the reference voltage of PECs of DERs to eliminate the effect of the harmonic voltage on the output voltage. At the same time, its value must be calculated for each frequency for compensation purposes.

There is a technique that uses a negative virtual harmonic impedance to compensate for the effect of transmission line impedance [143]. This makes it possible to share harmonic energy without deteriorating the power quality of consumer buses. Harmonic separation is achieved by reducing the transmission line impedance corresponding to a particular harmonic and by reducing the voltage that is added to the reference voltage.

Adaptive voltage droop control improves the quality of voltage regulation by eliminating the resistance between the voltage source and the point of common coupling [144]. In this case, the voltage drop across the resistance between the voltage source and the point of common coupling is included in the standard Q-U static control procedure. This method is highly effective at high loads in the microgrid, but at medium and low loads the effect of its use decreases.

Modified adaptive droop control is implemented using Equations (8) and (9):

$$\delta = -n_p P - n_i \int_{\infty}^t P d\tau - n_d \frac{dP}{dt},\tag{8}$$

$$U_0 = U - jQ - j_d \frac{dQ}{dt},\tag{9}$$

where δ —power angle and frequency calculated from power angle derivative; n_p , n_i and n_d —proportional, integral, and differential coefficients of output active power P; j and j_d —static and differential voltage droop coefficients; U_0 and U—specified (nominal) and regulated output voltage of PECs.

Modified adaptive droop control minimizes the transient circulating current between PECs of DERs, thereby improving the transient characteristic [145]. This provides the local microgrid controller with two degrees of freedom. Droop gain is used to control the magnitude of voltage and frequency, and transient coefficients are adjusted to actively suppress low-frequency power while it is shared among the PECs of DERs.

Droop is controlled by a proportional-integral-differential compensator that optimizes the frequency characteristic of the filter using three gain factors: proportional, integral, and differential. The first one is used for active power control, the second one is for power angle control together with PI reactive power compensator, and the third one is for the output voltage of PECs of DERs. Simulation results confirmed that when small signals are applied to the compensator input, they are amplified in the transient, thus improving the response of PECs of DERs.

When not using droop control, communication links between devices in the microgrid are required. To improve the efficiency of control, it is justified to adopt a centralized approach to control in microgrids as implemented in the following ways [71,146,147]:

 centralized control: local controllers transmit the data on the currents of all DERs to the central microgrid controller, which also monitors the voltage in the external distribution grid. The central controller calculates the contribution of each of the DERs to the total current, taking into account their specifications. The setpoints are also calculated from the output currents of DERs, which are transmitted from the central controller to local controllers. This approach allows for effective damping of transients but depends entirely on the reliability of the microgrid's information and communications technology infrastructure [148,149];

- "master-slave": the master PEC of the DER ensures that the microgrid maintains voltage and frequency within acceptable ranges, while the slave PECs of DERs either output or consume *P* and *Q*. This method is quite flexible, but its reliability is highly dependent on the proper operation of the master PEC of the DER under all load flows [150–152];
- voltage regulation and control of power allocation between PECs of DERs are performed by the central microgrid controller via low bandwidth communication links. In this case, local controllers provide harmonic and unbalance suppression [152–154].

Secondary control in microgrids is implemented through the EMS, which is responsible for maintaining power quality parameter values as well as frequency and voltage restoration when the primary control reserves prove insufficient [155,156]. The EMS provides synchronization of the microgrid with the external distribution grid, as well as optimal coordination of DERs in the microgrid. Secondary control can be implemented on the basis of centralized, distributed, and decentralized principles.

In centralized secondary control, the central microgrid controller sets the power values for local controllers [157]. Its main advantage is the online optimization of input parameters: setpoints, boundary conditions, network parameters, and predicted load flow.

As the number of DERs in a microgrid grows, the need to use distributed control methods arises. This allows for effective interaction between microgrid components to improve reliability, safety, and optimal performance [157]. In this case, the EMS communicates with the distributed control system by issuing its active and reactive power settings.

This is realized in a potential secondary control method that requires communication links. To this end, the central microgrid controller calculates the potential function for all PECs of DERs, minimizing its value to arrive at the specified power values.

$$\varphi_i(x_j) = \omega^{dg} \sum_{k=1}^{m_{dg}} p_k^{dg}(x_j) + \omega^{cn} \sum_{k=1}^{m_{cn}} p_k^{cn}(x_j) + \omega^{ob} p_k^{ob}(x_j),$$
(10)

where φ_i —potential function; x_j —measurement vector of the *j*-th element; p_k^{dg} , p_k^{cn} and p_k^{ob} —partial-potential measurement functions of the *k*-th component of the microgrid; ω^{dg} , ω^{cn} and ω^{ob} —weights of partial-potential functions.

The advantage of distributed control systems is that local controllers have more autonomy and can make decisions based on their interaction with each other. In distributed control, the control actions are formed taking into account both local load flow parameters and the microgrid operation mode as a whole [158].

Next, we consider the main categories of distributed control systems.

In agent-based systems, agents can be both physical and virtual entities, capable of responding to changes in the microgrid. Agents have intelligent algorithms that allow them to interact to solve common problems [159]. Agent-based systems use algorithms for microgrid state control, multicriteria assessment, and market competition [160]. The Java Agent Development Environment [161] is used to develop agents and ensure their coordination.

The predictive control method uses the forecast data on power consumption and generation in the microgrid, as well as the variables dependent on them [162]. This method allows one to solve multicriteria optimization problems using feedback mechanism predictions and controllable system constraints. The voltage prediction approach avoids voltage instability by controlling reactive power generation in the microgrid islanded mode.

A two-layer model of predictive control of hybrid energy complexes with renewable DERs makes microgrid operation more resilient to uncertainties [163]. When solving a problem with boundary variables, the reference settings of the on/off time of the GGS (DGS) are used, taking into account the deviations of the PV module power generation from the predicted values [164]. The calculation of the GGS (DGS) on/off time offset takes into account the microgrid state, load prediction data, and the current PV module power.

In distributed control systems, consensus-based methods can be applied to solve optimization problems to obtain a convergent solution for loading PECs of DERs. The multi-agent algorithm is implemented based on the average-consensus theorem for a group of agents that are designed to interact in the microgrid [165]. The application of the dynamic-consensus method optimizes the sharing of the negative sequence current under unbalanced load conditions.

Tertiary control is the slowest type of control, allowing for the coordination of several microgrids, so as to conform with the requirements of the external distribution grid [166,167]. When the specified values of active and reactive power are obtained from the DMS, control actions are formed within microgrids, taking into account the actual values of the generated power. The control of active and reactive power flows between microgrids and the external distribution grid contributes to solving the problem of the optimization of microgrid operation.

The application of intelligent control methods for power convert-based microgrids was discussed in a large number of studies [168,169]. One of the intelligent control methods is based on fuzzy inference, which includes:

- a fuzzifier—maps crisp (real-valued) input information into fuzzy information (fuzzy subset);
- an inference engine;
- a defuzzifier—converts the fuzzy output of the rule-based inference engine into crisp (real-valued) information to implement fuzzy inference control of the spatial vector of pulse-width modulation of the power converter. This provides the values of the magnitude and angle of the reference voltage vector needed to calculate switching moments and select switching algorithms.

The method of frequency control in primary regulation based on fuzzy inference is applicable to variable-speed wind turbines [170]. A fuzzy logic operator sets the pitch angle and power of the wind turbine inverter over the entire range of wind speeds. An advantage of this approach is that the primary power reserve is maintained independent of the current wind speed by controlling the pitch angle and torque of the generator, determined by fuzzy inference control. However, this does not take into account sudden load changes and possible damage in the microgrid [171]. The combination of fuzzy inference and swarm optimization was used to implement secondary frequency control in AC microgrids [172].

Another method of intelligent control is based on the use of artificial neural networks [173]. This method has been successfully applied to the estimation of load flow parameters, load forecasting, identification of islanding away of microgrids and switching control. Adaptive neuro-fuzzy ACS, for example, can be implemented in a frequency and voltage droop controller. This allows for the independence of ACS decision-making from the parameters of the power line and microgrid configuration, but it significantly increases the computational load [172].

Next, we list the features that are unique to ACS implementation in AC microgrids:

- selection of the type of ACS should be made on the basis of microgrid parameters (number of DERs, grid structure, etc.) and possible operating conditions;
- If ohmic resistance prevails in the power transmission line connecting the microgrid with the external distribution grid, then the ACS should use the static characteristic of voltage regulation as a function of active power (*U*-*P*) and the inverse static characteristic of frequency regulation as a function of reactive power (*f*-*Q*), which will provide the required quality of regulation;
- the ACS, which has a hierarchical structure, provides a more accurate distribution of active and reactive power between PECs of DERs, as well as the required quality of frequency and voltage regulation in microgrids;
- if there are unbalanced flows of active and reactive power in the transmission line connecting the microgrid with the external distribution grid, due to the presence of non-linear load in the microgrid, the ACS should employ state-of-the-art methods. These methods involve the following: applying a high-frequency signal, allocating

harmonic current between PECs of DERS connected in parallel without coupling between them, and adding a negative virtual harmonic impedance in order to allocate active and reactive power more accurately between PECs of DERs;

- control algorithms without communication links, which are based on frequency and voltage droop control, do not depend on the location of DERs and loads in the microgrid, but they are less efficient due to the lack of information exchange between PECs of DERs;
- ACS based on decentralized algorithms are increasingly being used because of the reduced risk of ACS failure due to damage to a single component, as opposed to centralized or agent-based ACSs.

4.2.2. Control Methods for Hybrid Microgrids

Hybrid microgrids have DC and AC networks connected to each other through a common inverter.

Refs. [14,21,24,61] listed the following challenges faced by hybrid microgrids:

- the ACS has a more complex structure and control algorithms due to the absence of a system-wide variable used for power allocation and frequency and voltage regulation;
- in the stand-alone, or islanded mode, power allocation between DC and AC microgrids cannot be ensured by droop-based algorithms;
- when there is a nonlinear load in the microgrid, it is necessary to ensure that the harmonic content power is distributed among the DERs;
- it is necessary to trade off the allocation of reactive power flows against voltage regulation at microgrid nodes under different operation conditions;
- droop control must not depend on the impedance of the power transmission line between the voltage source and the point of common coupling for optimal power allocation between DC and AC DERs;
- a hybrid microgrid requires the use of a reliable EMS to ensure reliability and the best performance;
- to provide online control of hybrid microgrid load flows, it is required to use hybrid ESSs for compensation of short-term unbalances of active power in the presence of pulsed loads.

Creating ACS and arranging the operation of hybrid microgrids is more complicated than it is in DC and AC microgrids. A hierarchical ACS of hybrid microgrids, as a rule, has three layers to implement the algorithms of primary, secondary, and tertiary control [174].

Centralized primary control is used in stand-alone hybrid microgrids with PV modules, wind turbines, and DGSs equipped with intelligent power regulators [175]. The regulators implement the maximum power point tracking (MPPT) function. In centralized ACSs, the central microgrid controller provides power balancing by determining the reference voltage and current values for local controllers. The power balancing algorithms are based on the principles of the grid following (the current loop controls the DC bus voltage) and source following (the current loop controls the active power of the grid, and the DC bus voltage is maintained by controlling the DERs). Experiments have confirmed the advantage of the source-following strategy [176]. In both cases, the algorithms of power allocation between DERs fail to take into account the presence of non-linear and unbalanced loads.

Applying the method of coordinated control of a hybrid microgrid with heterogeneous DERs equipped with multi-level power converters allows for taking into account the effect of a non-linear and unbalanced load. The DERs in this case are controlled by a multi-proportional resonant (MPR) controller.

Decentralized primary control of hybrid microgrids is realized by using a droop converter with the *I*-*U* characteristic for proportional power allocation between DC and AC converters operating in parallel [177]. The current control is implemented in the MPR controller, and the voltage control is implemented in the PI controller. The *I*-*U* droop coefficient is given by $V_{dc} = V_{dcref} - i_{dc}R_{vd}$, where V_{dc} —DC bus voltage; V_{dcref} —DC bus

voltage reference value; i_{dc} —DC bus current, R_{vd} —virtual impedance, which is maintained within the acceptable range according to Equation (11):

$$R_{vd} \leq \frac{\left[\left(V_{dcmax} - V_{dcmrg}\right) - \left(V_{dcmin} + V_{dcmrg}\right)\right]}{I_{dc(fl)}},\tag{11}$$

where V_{dcmax} —maximum voltage; V_{dcmin} —minimum voltage; V_{dcmrg} —design voltage reserve; $I_{dc(fl)}$ —full-load current on the DC side.

This approach does not take into account the AC grid voltage droop control of the hybrid microgrid, which is used in implementing bi-directional power droop control based on normalized and general proportional power allocation [178] (Figure 9).



Figure 9. Bi-directional static *P-f* control of a hybrid microgrid.

The normalization technique is used to combine the static characteristics of AC and DC grids that use different values. Thus, the normalized frequency in the AC grid and the DC bus voltage are calculated as per Equations (12) and (13).

$$Nf = \frac{f - f^*}{(f_{max} - f_{min})/2};$$
(12)

$$NV_{dc} = \frac{V - V_{dc}^*}{(V_{max} - V_{min})/2},$$
(13)

where *Nf* and *NV*_{pu}—normalized values of frequency in the AC grid and DC bus voltage; $f^* = (f_{max} + f_{min})/2$ —nominal frequency value; $V_{dc}^* = (V_{max} + V_{min})/2$; f_{max} and f_{min} —maximum and minimum frequency in the AC grid; V_{max} and V_{min} —maximum and minimum voltages in the DC grid.

The two-stage bi-directional droop control method in a hybrid microgrid for optimal power allocation is based on the application of the AC-DC hybrid droop controller [179]. The principle of setting droop in a hybrid microgrid is presented in Figure 10.



Figure 10. The principle of setting droop in a hybrid microgrid.

The hybrid droop controller uses the V_{dc} -P and ω -P characteristics implemented in the intermediate converter. At the beginning of the operation, the hybrid controller of microgrids gets the initial operating conditions from the AC and DC microgrids (indicated by the black dots in Figure 10). According to the droop control principle, achieving a proportional power allocation between AC and DC microgrids means $\omega_{ac,pu} = V_{dc,pu} = \xi_s$ (the red horizontal line in Figure 10). Thus, as the AC and DC loads in the microgrids change, the value of the active power $P_{MC, t_{k+1}}$ to be allocated will update according to the following rule:

$$P_{MC, t_{k+1}} = \frac{1}{k_{ac} + k_{dc}} \Big(V_{dc, pu, t_{k+1}} - \omega_{ac, pu, t_{k+1}} \Big), \ k \ge 0,$$
(14)

where k_{ac} and k_{dc} —droop coefficients for AC and DC microgrids, respectively; $V_{dc,pu,t_{k+1}}$ —measured DC voltage at time $t = t_{k+1}$; $\omega_{ac,pu,t_{k+1}}$ —measured AC frequency at time $t = t_{k+1}$. Centralized secondary control is implemented through:

- supervisory control based on optimization algorithms;
- coordinated control of DC and AC grids as part of a hybrid microgrid;
- intelligent supervisory control [180].

Supervisory control based on optimization algorithms makes it possible to maximize the use of renewable DERs and minimize the load of DGSs. This takes into account the forecasting error of electricity generation of renewable DERs and its intermittent nature. This extends the battery life of the ESS battery storage by optimizing the charging/discharging processes, as well as reduces the load of the common inverter between the DC and AC grids [181].

Coordinated control allows for efficient control of the hybrid microgrid in both gridconnected and islanded modes of operation. At the same time, the ACS provides MPPT operation of wind turbines and PV modules in the grid-connected mode of microgrid operation [182,183]. In the islanded mode, on-MPPT and off-MPPT modes can be used by wind turbines and PV modules, depending on the control actions of the power control system.

Intelligent supervisory control is implemented by a fuzzy controller that uses a set of rules to control the operation of DC and AC grids as part of a hybrid microgrid [184]. Known practical solutions for EMS implementation make it possible to effectively solve optimization problems aimed at minimizing operating costs in microgrids by means of intelligent charging/discharging control of ESSs [185]. Moreover, the fuzzy controller responds even to small changes in power consumption in hybrid microgrids by adjusting the setpoints for the amount of power output by the fuel-fired DERs.

Decentralized secondary control in hybrid microgrids is effective when distributed ESSs are available, which allows for the implementation of algorithms for local, systemwide, and individual allocation of ESS power [186]. In this case, the central microgrid controller is replaced by a multilevel ACS.

Hierarchical coordinated control is based on the agent-based approach. To this end, the hybrid microgrid is modeled using a differential hybrid Petri net [187,188]. Switching control is implemented at local and system-wide levels with voltage control, including in transients associated with changes in microgrid load flows under emergency disturbances.

Decentralized secondary control in hybrid microgrids can be implemented based on the modified droop for five operating states: ESS charging/discharging, limited ESS charging, limited PV module power, switched off ESSs, and limited power output delivered to the external distribution grid [189,190].

Distributed secondary control can be implemented with communication between PECs of DERs (local controllers) for frequency and voltage regulation in microgrids.

Hierarchical control with a multi-agent structure and consensus-based economic scheduling was proposed and implemented through a case study of a reconfigurable hybrid microgrid.

The agents are:

- energy service company—ensures the safe and reliable operation of the external distribution grid;
- microgrid: responsible for monitoring, economic scheduling, and management of DERs within the microgrid;
- DER: responsible for monitoring, protecting, and implementing primary control functions of each DER [191].

In stand-alone hybrid microgrids, a two-layer multi-agent frequency regulation structure based on an optimized average-consensus algorithm can be implemented in the AC grid. Agents manage both generating DERs and loads. The ACS implements a multi-stage load shedding to restore the frequency in the microgrid when the primary control reserve is exhausted. The advantage of this approach is the efficient exchange of information, as well as quick decision-making when the circuit topology and operating conditions change.

Distributed coordinated control of renewable DERs using subgradients is based on a multi-agent approach. Renewable DERs exchange information with neighboring agents and receive frequency information from the microgrid. Renewable DERs implement bilevel control, with the upper level predicting the amount of active power generation and the lower level controlling the active power. Fuel-fired generating units act as auxiliary sources to regulate the voltage in microgrids, provided that the load profile is served by the output of renewable DERs. When there is a shortage of power from renewable DERs, fuel-fired generating plants serve as the main sources. Frequency measurements are used to set the amount of generation by the fuel-fired generation units, and information about the generation of adjacent DERs is used to obtain information about the amount of the active power reserve [192]. Agent-based control facilitates the reliability of hybrid microgrids through parallel computing. Since communication between agents is prone to security threats, it requires the implementation of measures to ensure information security [193].

Hybrid microgrids have unique features that make them stand apart with respect to the principles of ACS design to be conformed to:

- reliable and efficient management of power flows requires the implementation of a sophisticated control strategy [194];
- they require an additional intermediate converter between the DC and AC grid, which is necessary to maintain the balance of power in the microgrid, both in grid-connected and islanded modes [195];

• the absence of a system-wide variable used for power allocation and frequency and voltage regulation necessitates the use of an ACS with a more elaborate structure and control algorithms [196].

A new trend in the design of modern ACSs of hybrid microgrids is the use of hybrid AC-DC droop, as well as unified frequency and voltage control on both sides (DC and AC grids) [197,198].

5. Experimental Microgrids and Test Systems

Microgrids have a wide geographic range of applications, which is due to innovative research and development of various components for microgrid control systems. To this end, experimental microgrids were built for the specific purpose of determining their design criteria and evaluating the performance of their specifications during operation. The results of research in the field of microgrid testing were reported in [56], where the main recommendations for the design, operation, and integration of microgrids in distribution networks were presented. Microgrid testing can be performed with a demonstration microgrid or a laboratory microgrid emulator. However, laboratory microgrid emulators used for testing, despite the relatively low cost of their design and development, are not very popular [199].

At present, there are two basic concepts for the design of microgrid control systems: the American one, developed by CERTS [53], and the European one, implemented in the MICROGRIDS and MORE MICROGRIDS projects [200]. The key difference between them is that the first concept addresses the issues of heating in addition to the issues of power supply. Other microgrids projects are known to have been implemented around the world:

- NEDO (India [201]): adoption of microgrids at the national level using centralized control systems implemented in the context of poor information and communication infrastructure (TWACS and GPRS communication technologies were used);
- Microgrid UW-Madison [202]: a study of modeling and control problems of microgrids containing generating units connected via PECs in the case of integrating diesel gensets into them;
- DISPOWER, Am Steinweg [203]: a study of the operation of microgrids in a residential complex using a Power Operation and Power Quality Management System (PoMS). The PoMS implements optimization algorithms to control the operation of microgrids, controllers of DERs, and demand response. A key defining feature of the PoMS is a central unit and several decentralized control units;
- Kythnos island microgrid (Greece) [70]: testing of a decentralized approach to microgrid control. An agent-based load controller was used to monitor the state of the island's microgrid by making voltage, current, and frequency measurements. The measurement data were used to coordinate power consumption management in the island's microgrid;
- University of Manchester microgrid/flywheel energy storage laboratory prototype [204]: a study of the joint operation of a synchronous generator and an induction motor coupled together and acting as a single electrical installation, as well as a flywheel storage unit integrated into the microgrid through an inverter. A unique feature of the above topology is the connection of the flywheel storage by means of two inverter units through a line reactance and a coupling transformer. The flywheel storage forms the reference voltage and frequency in the microgrid and provides the necessary fault current for the protection devices to trip.

To maintain the voltage and frequency in experimental and test microgrid circuits, different approaches are used to implement their control systems: centralized control [70,201,204], decentralized control [53,202], and distributed control based on multi-agent systems [200,203]. References [205,206] provided generalized conclusions on the approaches that are deemed preferable in microgrid control systems. Asian countries use centralized control, in contrast to North American countries where decentralized control is employed, and European countries use both centralized and distributed control based on multi-agent systems.

6. Discussion

Microgrid design technology studies are one of the key areas in the expansion planning of power grids, as they make it possible to improve the reliability of power supply to consumers and power quality without significant investment outlays. The highly dispersed integration of diversified DERs poses challenges to the implementation of microgrid control systems.

The availability of different approaches to building microgrid control systems (centralized, decentralized, and distributed) [70] inhibits the formation of a holistic perception of this field, as many contributions try to improve on previously developed solutions, for example, by adding new ways of collecting and processing information. The presence of a hierarchical control system, as the single possible control arrangement [70,98], comes down to a number of variations in its implementation. Therefore, the choice of an approach to the arrangement of microgrid control comes down to the state of the information and communications technology infrastructure. However, when it is poorly developed [201], a centralized approach to microgrid control is implemented. This streamlines the process of interaction with the power system, given that this technology is the most mature at present. If there is no technical possibility to implement centralized control of microgrids, decentralized approaches based on droop control [132,133] and its modifications [134,135,144,145] are applied. A number of studies showed interest in applying algorithms that use virtual impedance [137,138] and virtual inertia [139,140].

Extra difficulties arise when building hybrid microgrids, in which the basic known control algorithms cannot be implemented due to the lack of frequency and certain peculiarities of DC network parameters [172], so one has to rely on two-parameter algorithms [178].

7. Recommendations and Future Research

When choosing an approach to building a microgrid control system, one has to solve a multi-parameter problem, which does not lend itself to the treatment based on conventional criteria (simplicity, convenience, speed). The choice of the microgrid control system should be considered from the standpoint of holonic architectures [103], where each microgrid component is both a stand-alone unit and a part of the microgrid. This principle is well-aligned with the distributed approach to control.

To implement holarchic control [104] on the basis of the distributed approach it is necessary to create digital twins of components and microgrids as a whole. Their use in metaheuristic algorithms will form both predictive and look-ahead controls. The development of quantum computing [207] may allow this concept to be implemented in future research in this field.

8. Conclusions

In the context of large-scale development of DERs, in order to ensure the ability to control power grids of distribution grids, microgrids should be formed so as to mitigate the challenges presented by the intermittent power generation by DERs and consumption by heterogeneous loads.

The optimal choice of structure for microgrids (DC, AC, or AC-DC) reduces power losses and grid modernization costs, ensures efficient frequency and voltage regulation, and maintains power quality parameters within required boundaries, as well as increases the reliability of power supply to consumers.

Microgrids stand apart in that it is impossible to design a standardized ACS structure, which is due to the great variety of DERs, microgrid structures, types of electrical loads, and other factors. Therefore, the ACS structure is formed on a case-by-case basis for each microgrid during its design, taking into account the pros and cons of possible options.

It is possible to ensure the reliable operation of DERs and electrical loads of consumers in the grid-connected and islanded modes of microgrid operation only if the choice of the structure of DERs is optimal. This article analyzed the state-of-the-art principles of microgrid design that influence the choice of microgrid power flow control methods, as well as power flow control methods themselves. The analysis will prove instrumental in streamlining the process of creating ACSs of microgrids in existing distribution grids of low-voltage AC and AC-DC microgrids.

Author Contributions: Conceptualization, P.I. and V.V.; methodology, P.I. and S.F.; software, V.V.; validation, P.I.; formal analysis, K.S.; investigation, V.V., P.I. and K.S.; resources, S.F.; data curation, V.V. and S.F.; writing—original draft preparation, P.I. and V.V.; writing—review and editing, S.F. and K.S.; visualization, V.V.; supervision, K.S.; project administration, P.I.; funding acquisition, S.F. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Data Availability Statement: Data sharing not applicable. No new data were created or analyzed in this study. Data sharing is not applicable to this article.

Conflicts of Interest: The funders had no role in the design of the study; in the collection, analyses, or interpretation of data; in the writing of the manuscript, or in the decision to publish the results.

Abbreviations

DER	distributed energy resources
RES	renewable energy sources
ESS	energy storage systems
FC	fuel cell
GGS	gas generator sets
DGS	diesel generator sets
ACS	automatic control systems
PCC	point of common coupling
IC	interlinking converter
PEC	power electronic converters
DMS	distribution management system
PI	proportional-integral
PR	proportional-resonance
MPPT	maximum power point tracking
MPR	multi-proportional resonant
VSM	virtual synchronous machine
PoMS	power flow and power quality management system

References

- Filippov, S.P.; Dilman, M.D.; Ilyushin, P.V. Distributed Generation of Electricity and Sustainable Regional Growth. *Therm. Eng.* 2019, *66*, 869–880. [CrossRef]
- Pandiyan, P.; Sitharthan, R.; Saravanan, S.; Prabaharan, N.; Ramji Tiwari, M.; Chinnadurai, T.; Yuvaraj, T.; Devabalaji, K.R. A comprehensive review of the prospects for rural electrification using stand-alone and hybrid energy technologies. *Sustain. Energy Technol. Assess.* 2022, 52, 102155. [CrossRef]
- 3. Zhang, X.; Chan, S.H.; Ho, H.K.; Tan, S.C.; Li, M.; Li, G.; Li, J.; Feng, Z. Towards a smart energy network: The roles of fuel/electrolysis cells and technological perspectives. *Int. J. Hydrogen Energy* **2015**, *40*, 6866–6919. [CrossRef]
- Jamal, T.; Salehin, S. Hybrid renewable energy sources power systems. *Hybrid Renew. Energy Syst. Microgrids* 2021, 179–214. [CrossRef]
- Cagnano, A.; de Tuglie, E.; Mancarella, P. Microgrids: Overview and guidelines for practical implementations and operation. *Appl. Energy* 2020, 258, 114039. [CrossRef]
- 6. Li, R. Protection and control technologies of connecting to the grid for distributed power resources. *Distrib. Power Resour.* 2019, 121–144. [CrossRef]
- Ilyushin, P.; Volnyi, V.; Suslov, K.; Filippov, S. Review of Methods for Addressing Challenging Issues in the Operation of Protection Devices in Microgrids with Voltages of up to 1 kV That Integrates Distributed Energy Resources. *Energies* 2022, 15, 9186. [CrossRef]
- 8. Zheng, D.; Zhang, W.; Netsanet Alemu, S.; Wang, P.; Bitew, G.T.; Wei, D.; Yue, J. Key technical challenges in protection and control of microgrid. *Microgrid Prot. Control* **2021**, 45–56. [CrossRef]

- 9. Ku Ahmad, K.N.E.; Selvaraj, J.; Rahim, N.A. A review of the islanding detection methods in grid-connected PV inverters. *Renew.* Sustain. Energy Rev. 2013, 21, 756–766. [CrossRef]
- Ilyushin, P.V.; Pazderin, A.V. Requirements for power stations islanding automation an influence of power grid parameters and loads. In Proceedings of the 2018 International Conference on Industrial Engineering, Applications and Manufacturing (ICIEAM), Moscow, Russia, 15–18 May 2018.
- 11. Ilyushin, P.V.; Suslov, K.V. Operation of automatic transfer switches in the networks with distributed generation. In Proceedings of the 2019 IEEE Milan PowerTech, Milan, Italy, 23–27 June 2019. [CrossRef]
- 12. Zheng, D.; Zhang, W.; Alemu, S.N.; Wang, P.; Bitew, G.T.; Wei, D.; Yue, J. Microgrid Protection and Control. In *Microgrid Protection and Control*; Elsevier: Amsterdam, The Netherlands, 2021. [CrossRef]
- 13. Justo, J.J.; Mwasilu, F.; Lee, J.; Jung, J.W. AC-microgrids versus DC-microgrids with distributed energy resources: A review. *Renew. Sustain. Energy Rev.* 2013, 24, 387–405. [CrossRef]
- Sahoo, S.K.; Sinha, A.K.; Kishore, N.K. Control Techniques in AC, DC, and Hybrid AC-DC Microgrid: A Review. *IEEE J. Emerg.* Sel. Top. Power Electron. 2018, 6, 738–759. [CrossRef]
- 15. Planas, E.; Andreu, J.; Gárate, J.I.; Martínez De Alegría, I.; Ibarra, E. AC and DC technology in microgrids: A review. *Renew. Sustain. Energy Rev.* 2015, 43, 726–749. [CrossRef]
- Pasonen, R. Community Microgrid—A Building Block of Finnish Smart Grid. Master's Thesis, TUT, Tampere, Finland, 2010; 119p.
 Cornélusse, B.; Savelli, I.; Paoletti, S.; Giannitrapani, A.; Vicino, A. A community microgrid architecture with an internal local market. *Appl. Energy* 2019, 242, 547–560. [CrossRef]
- 18. Shehadeh, H.; Siam, J.; Abdo, A. Enhancement of a microgrid operation in blackouts using central control scheme and network reconfiguration: A case study. *Electr. Power Syst. Res.* **2022**, 212, 108632. [CrossRef]
- 19. Møller Sneum, D. Barriers to flexibility in the district energy-electricity system interface—A taxonomy. *Renew. Sustain. Energy Rev.* 2021, 145, 111007. [CrossRef]
- 20. Sen, S.; Kumar, V. Microgrid control: A comprehensive survey. Annu. Rev. Control 2018, 45, 118–151. [CrossRef]
- Pourbehzadi, M.; Niknam, T.; Aghaei, J.; Mokryani, G.; Shafie-khah, M.; Catalão, J.P.S. Optimal operation of hybrid AC/DC microgrids under uncertainty of renewable energy resources: A comprehensive review. *Int. J. Electr. Power Energy Syst.* 2019, 109, 139–159. [CrossRef]
- 22. Rokrok, E.; Shafie-khah, M.; Catalão, J.P.S. Review of primary voltage and frequency control methods for inverter-based islanded microgrids with distributed generation. *Renew. Sustain. Energy Rev.* **2018**, *82*, 3225–3235. [CrossRef]
- Murray, W.; Adonis, M.; Raji, A. Voltage control in future electrical distribution networks. *Renew. Sustain. Energy Rev.* 2021, 146, 111100. [CrossRef]
- 24. Jain, D.; Saxena, D. Comprehensive review on control schemes and stability investigation of hybrid AC-DC microgrid. *Electr. Power Syst. Res.* 2023, *218*, 109182. [CrossRef]
- 25. Wang, C.; Nehrir, H.; Lin, F.; Zhao, J. From hybrid energy systems to microgrids: Hybridization techniques, configuration, and control. In Proceedings of the IEEE PES General Meeting, Minneapolis, MN, USA, 25–29 July 2010. [CrossRef]
- 26. Reveron Baecker, B.; Candas, S. Co-optimizing transmission and active distribution grids to assess demand-side flexibilities of a carbon-neutral German energy system. *Renew. Sustain. Energy Rev.* **2022**, *163*, 112422. [CrossRef]
- Wang, C.; Nehrir, M.H. Power management of a stand-alone wind/photovoltaic/fuel cell energy system. *IEEE Trans. Energy Convers.* 2008, 23, 957–967. [CrossRef]
- 28. Sood, V.K.; Abdelgawad, H. Microgrids architectures. Distrib. Energy Resour. Microgrids Integr. Chall. Optim. 2019, 1–31. [CrossRef]
- Natesan, C.; Ajithan, S.K.; Palani, P.; Kandhasamy, P. Survey on Microgrid: Power Quality Improvement Techniques. ISRN Renew. Energy 2014, 2014, 342019. [CrossRef]
- 30. Hatziargyriou, N.; Asano, H.; Iravani, R.; Marnay, C. Microgrids. IEEE Power Energy Mag. 2007, 5, 78–94. [CrossRef]
- Muljadi, E.; Samaan, N.; Gevorgian, V.; Li, J.; Pasupulati, S. Short circuit current contribution for different wind turbine generator types. In Proceedings of the Power and Energy Society General Meeting, Minneapolis, MN, USA, 25–29 July 2010; IEEE: Piscataway, NJ, USA, 2010; pp. 1–8.
- Fischer, M.; Mendonca, A. Representation of variable speed full conversion Wind Energy Converters for steady state short-circuit calculations. In Proceedings of the 2011 IEEE Power and Energy Society General Meeting, Detroit, MI, USA, 24–28 July 2011; pp. 1–7. [CrossRef]
- 33. Sivakumar, K.; Jayashree, R.; Danasagaran, K. Efficiency-driven planning for sizing of distributed generators and optimal construction of a cluster of microgrids. *Eng. Sci. Technol. Int. J.* **2021**, *24*, 1153–1167. [CrossRef]
- Wu, D.; Ma, X.; Huang, S.; Fu, T.; Balducci, P. Stochastic optimal sizing of distributed energy resources for a cost-effective and resilient Microgrid. *Energy* 2020, 198, 117284. [CrossRef]
- 35. Akram, U.; Khalid, M.; Shafiq, S. Optimal sizing of a wind/solar/battery hybrid grid-connected microgrid system. *IET Renew. Power Gener.* **2018**, *12*, 72–80. [CrossRef]
- Hajian, M.; Rosehart, W.D.; Zareipour, H. Probabilistic power flow by Monte Carlo simulation with Latin supercube sampling. IEEE Trans. Power Syst. 2013, 28, 1550–1559. [CrossRef]
- Silva, E.N.M.; Rodrigues, A.B.; da Guia da Silva, M. Approximated and Iterative Power Flow Algorithms for Islanded DC Microgrids. *Electr. Power Syst. Res.* 2023, 215, 108972. [CrossRef]

- Ma, Q.; Huang, X.; Wang, F.; Xu, C.; Babaei, R.; Ahmadian, H. Optimal sizing and feasibility analysis of grid-isolated renewable hybrid microgrids: Effects of energy management controllers. *Energy* 2022, 240, 122503. [CrossRef]
- Elkadeem, M.R.; Wang, S.; Sharshir, S.W.; Atia, E.G. Feasibility analysis and techno-economic design of grid-isolated hybrid renewable energy system for electrification of agriculture and irrigation area: A case study in Dongola, Sudan. *Energy Convers. Manag.* 2019, 196, 1453–1478. [CrossRef]
- Fioriti, D.; Lutzemberger, G.; Poli, D.; Duenas-Martinez, P.; Micangeli, A. Coupling economic multi-objective optimization and multiple design options: A business-oriented approach to size an off-grid hybrid microgrid. *Int. J. Electr. Power Energy Syst.* 2021, 127, 106686. [CrossRef]
- Xie, P.; Jia, Y.; Lyu, C.; Wang, H.; Shi, M.; Chen, H. Optimal sizing of renewables and battery systems for hybrid AC/DC microgrids based on variability management. *Appl. Energy* 2022, 321, 119250. [CrossRef]
- 42. Swaminathan, S.; Pavlak, G.S.; Freihaut, J. Sizing and dispatch of an islanded microgrid with energy flexible buildings. *Appl. Energy* **2020**, *276*, 115355. [CrossRef]
- 43. Huang, W.; Sun, C.H.; Wu, Z.P.; Zhang, J.H. A review on micro-grid technology containing distributed generation system. *Power Syst. Technol.* **2009**, *9*, 14–18.
- 44. Hong, Y.Y.; Wei, Y.H.; Chang, Y.R.; Lee, Y.D.; Liu, P.W. Fault Detection and Location by Static Switches in Microgrids Using Wavelet Transform and Adaptive Network-Based Fuzzy Inference System. *Energies* **2014**, *7*, 2658–2675. [CrossRef]
- Kroposki, B.; Pink, C.; Lynch, J.; John, V.; Daniel, S.M.; Benedict, E.; Vihinen, I. Development of a high-speed static switch for distributed energy and microgrid applications. In Proceedings of the Fourth Power Conversion Conference-NAGOYA, PCC-NAGOYA 2007—Conference Proceedings, Nagoya, Japan, 2–5 April 2007; pp. 1418–1423. [CrossRef]
- Prakash, K.; Islam, F.R.; Mamun, K.A.; Lallu, A.; Cirrincione, M. Reliability of Power Distribution Networks with Renewable Energy Sources. In Proceedings of the 2017 4th Asia-Pacific World Congress on Computer Science and Engineering, APWC on CSE 2017, Mana Island, Fiji, 11–13 December 2017; pp. 187–192. [CrossRef]
- Mittal, A.; Rajput, A.; Johar, K.; Kandari, R. Microgrids, their types, and applications. *Microgrids Model. Control Appl.* 2022, 3–40. [CrossRef]
- Sakis Meliopoulos, A.P. Challenges in simulation and design of μGrids. In Proceedings of the IEEE Power Engineering Society Transmission and Distribution Conference, Chicago, IL, USA, 21–25 July 2002; Volume 1, pp. 309–314. [CrossRef]
- 49. Gabbar, H.A. Smart energy grid infrastructures and interconnected micro energy grids. *Smart Energy Grid Eng.* **2017**, 23–45. [CrossRef]
- 50. IEEE SA—IEEE 1709–2018. (n.d.). Available online: https://standards.ieee.org/ieee/1709/7073/ (accessed on 14 January 2023).
- 51. Mirsaeidi, S.; Mat Said, D.; Wazir Mustafa, M.; Hafiz Habibuddin, M.; Ghaffari, K. Progress and problems in micro-grid protection schemes. *Renew. Sustain. Energy Rev.* 2014, *37*, 834–839. [CrossRef]
- 52. Lin, X.; Zhang, R.; Tong, N.; Li, X.; Li, M.; Yang, D. Regional protection scheme designed for low-voltage micro-grids. *Int. J. Electr. Power Energy Syst.* **2015**, *64*, 526–535. [CrossRef]
- The CERTS Microgrid Concept, as Demonstrated at the CERTS/AEP Microgrid Test Bed. (n.d.). Available online: https: //www.researchgate.net/publication/328600356_The_CERTS_Microgrid_Concept_as_Demonstrated_at_the_CERTSAEP_ Microgrid_Test_Bed (accessed on 4 January 2023).
- 54. Arabali, A.; Ghofrani, M.; Bassett, J.B.; Pham, M.; Moeini-Aghtaei, M. Optimum Sizing and Siting of Renewable-Energy-based DG Units in Distribution Systems. *Optim. Renew. Energy Syst. Recent Perspect.* **2017**, 233–277. [CrossRef]
- Patrao, I.; Figueres, E.; Garcerá, G.; González-Medina, R. Microgrid architectures for low voltage distributed generation. *Renew. Sustain. Energy Rev.* 2015, 43, 415–424. [CrossRef]
- 56. *IEEE STD 1547-2018*; IEEE Standard for Interconnection and Interoperability of Distributed Energy Resources with Associated Electric Power Systems Interfaces. IEEE: Manhattan, NY, USA, 2018; pp. 1–138.
- Prodanović, M.; Green, T.C. High-quality power generation through distributed control of a power park microgrid. *IEEE Trans. Ind. Electron.* 2006, 53, 1471–1482. [CrossRef]
- Ilyushin, P.V. The analysis of dispersed generation influence on power system automatics settings and function algorithms. In Proceedings of the Methodological Problems in Reliability Study of Large Energy Systems (RSES), E3S Web of Conferences, Irkutsk, Russia, 2–7 July 2018; Volume 58, p. 02001. [CrossRef]
- Mehta, S.; Puri, V. A review of different multi-level inverter topologies for grid integration of solar photovoltaic system. *Renew.* Energy Focus 2022, 43, 263–276. [CrossRef]
- 60. Lidula, N.W.A.; Rajapakse, A.D. Microgrids research: A review of experimental microgrids and test systems. *Renew. Sustain. Energy Rev.* **2011**, *15*, 186–202. [CrossRef]
- 61. Jia, L.; Zhu, Y.; Wang, Y. Architecture design for new AC-DC hybrid micro-grid. In Proceedings of the 2015 IEEE 1st International Conference on Direct Current Microgrids, ICDCM 2015, Atlanta, GA, USA, 7–10 June 2015; pp. 113–118. [CrossRef]
- 62. Zolfaghari, M.; Gharehpetian, G.B.; Shafie-khah, M.; Catalão, J.P.S. Comprehensive review on the strategies for controlling the interconnection of AC and DC microgrids. *Int. J. Electr. Power Energy Syst.* **2022**, *136*, 107742. [CrossRef]
- 63. Zhang, X.; Lin, F.; Ma, H.; Zhao, B.; Huang, J. The Proposed Two-Stage Parameter Design Methodology of a Generalized Resonant DC Transformer in Hybrid AC/DC Microgrid with Optimum Active Power Transmission. In *Holistic Design of Resonant DC Transformer on Constant Voltage Conversion, Cascaded Stability and High Efficiency;* Springer: Singapore, 2023; pp. 83–113. [CrossRef]

- 64. CIGRE Green Books | Book Titles in This Series. (n.d.). Available online: https://www.springer.com/series/15209/books (accessed on 13 January 2023).
- 65. Ilyushin, P.V.; Sukhanov, O.A. The Structure of Emergency-Management Systems of Distribution Networks in Large Cities. *Russ. Electr. Eng.* **2014**, *85*, 133–137. [CrossRef]
- Sinsel, S.R.; Riemke, R.L.; Hoffmann, V.H. Challenges and solution technologies for the integration of variable renewable energy sources—A review. *Renew. Energy* 2020, 145, 2271–2285. [CrossRef]
- 67. Kottayil, S.K. (Ed.) Smart Microgrids, 1st ed.; CRC Press: Boca Raton, FL, USA, 2020. [CrossRef]
- Ortiz, L.; González, J.W.; Gutierrez, L.B.; Llanes-Santiago, O. A review on control and fault-tolerant control systems of AC/DC microgrids. *Heliyon* 2020, 6, e04799. [CrossRef]
- 69. El Bassam, N. Grid challenges: Integration of distributed renewables with the national grid. In *Distributed Renewable Energies for Off-Grid Communities: Empowering a Sustainable, Competitive, and Secure Twenty-First Century;* Elsevier: Amsterdam, The Netherlands, 2021; pp. 451–456. [CrossRef]
- Hatziargyriou, N. (n.d.). Microgrids: Architectures and Control. Available online: https://books.google.com/books/about/ Microgrids.html?hl=ru&id=ywxzAgAAQBA (accessed on 14 January 2023).
- Olivares, D.E.; Mehrizi-Sani, A.; Etemadi, A.H.; Cañizares, C.A.; Iravani, R.; Kazerani, M.; Hajimiragha, A.H.; Gomis-Bellmunt, O.; Saeedifard, A.; Palma-Behnke, R.; et al. Trends in microgrid control. *IEEE Trans. Smart Grid* 2014, *5*, 1905–1919. [CrossRef]
- Bordons, C.; Garcia-Torres, F.; Ridao, M.A. Model Predictive Control of Microgrids; Springer: Berlin/Heidelberg, Germany, 2020. [CrossRef]
- Marija, D.I. Toward a Unified Modeling and Control for Sustainable and Resilient Electric Energy System. Found. Trends®Electr. Energy Syst. 2016, 1, 1–141. [CrossRef]
- 74. Energies | Special Issue: Communications in Microgrids. (n.d.). Available online: https://www.mdpi.com/journal/energies/special_issues/Communications_in_Microgrids?view=default&listby=type (accessed on 13 January 2023).
- 75. Van den Broeck, G.; Stuyts, J.; Driesen, J. A critical review of power quality standards and definitions applied to DC microgrids. *Appl. Energy* **2018**, 229, 281–288. [CrossRef]
- International Electrotechnical Commission. (n.d.). Low voltage Electrical Installations. Part 1 : Fundamental Principles, Assessment of General Characteristics, Definitions. 93. Available online: https://standards.iteh.ai/catalog/standards/iec/51c4 126e-8510-4459-bbc9-27b5bfca80f8/iec-60364-1 (accessed on 12 January 2023).
- IEC 61000-4-30:2015+AMD1:2021 CSV | IEC Webstore. (n.d.). IEEE 115 [115-2019—IEEE Guide for Test Procedures for Synchronous Machines Including Acceptance and Performance Testing and Parameter Determination for Dynamic Analysis. 2020. Available online: https://webstore.iec.ch/publication/68642]/ (accessed on 12 January 2023).
- IEC 61508:2010 CMV | IEC Webstore. (n.d.). Available online: https://webstore.iec.ch/publication/22273 (accessed on 14 January 2023).
- IEC 61499-1:2012 | IEC Webstore. (n.d.). Available online: https://webstore.iec.ch/publication/5506&preview=1 (accessed on 14 January 2023).
- IEC 61499-2:2012 | IEC Webstore. (n.d.). Available online: https://webstore.iec.ch/publication/5507 (accessed on 14 January 2023).
- IEC 61499-4:2013 | IEC Webstore. (n.d.). Available online: https://webstore.iec.ch/publication/5508 (accessed on 14 January 2023).
- IEC 61970:2023 SER | IEC Webstore | Automation, Cyber Security, Smart City, Smart Energy, Smart Grid, CGMES. (n.d.). Available online: https://webstore.iec.ch/publication/61167 (accessed on 14 January 2023).
- IEC Smart Grid Standardization Roadmap. International Electrotechnical Commission (IEC), Tech. Rep. Ed. 1.0, 2010. SMB Smart Grid Strategic Group (SG3). 2010; 136p. Available online: https://www.smartgrid.gov/files/documents/IEC_Smart_Grid_ Standardization_Roadmap_201005.pdf (accessed on 14 January 2023).
- Tissue Overview of All Parts | IEC 61850 Tissue Database. (n.d.). Available online: https://iec61850.tissue-db.com/parts.mspx (accessed on 12 January 2023).
- 85. IEC 62351:2023 SER | IEC Webstore | Cyber Security, Smart City. (n.d.). Available online: https://webstore.iec.ch/publication/6912 (accessed on 14 January 2023).
- IEC TR 62357-1:2016 | IEC Webstore | Smart Energy, Smart Grid. (n.d.). Available online: https://webstore.iec.ch/publication/26 251 (accessed on 14 January 2023).
- 87. IEC TR 62357-2:2019 | IEC Webstore. (n.d.). Available online: https://webstore.iec.ch/publication/28523 (accessed on 14 January 2023).
- IEEE Std P2030 | IEEE Standards Coordinating Committee 21 (SCC21). (n.d.). Available online: https://sagroups.ieee.org/scc21/ standards/ieee-std-2030-2011/ (accessed on 14 January 2023).
- IEEE SA—IEEE 2030.10-2021. (n.d.). Available online: https://standards.ieee.org/ieee/2030.10/10742/ (accessed on 14 January 2023).
- ISO—ISO 52016-1:2017—Energy Performance of Buildings—Energy Needs for Heating and Cooling, Internal Temperatures and Sensible and Latent Heat Loads—Part 1: Calculation Procedures. (n.d.). Available online: https://www.iso.org/standard/65696. html (accessed on 14 January 2023).

- 91. IEC 61851-23:2014 | IEC Webstore | LVDC. (n.d.). Available online: https://webstore.iec.ch/publication/6032 (accessed on 14 January 2023).
- Suslov, K.; Shushpanov, I.; Buryanina, N.; Ilyushin, P. Flexible power distribution networks: New opportunities and applications. In Proceedings of the 9th International Conference on Smart Cities and Green ICT Systems (SMARTGREENS), Prague, Czech Republic, 2–4 May 2020; pp. 57–64. [CrossRef]
- 93. Khemmook, P.; Prompinit, K.; Surinkaew, T. Control of a microgrid using robust data-driven-based controllers of distributed electric vehicles. *Electr. Power Syst. Res.* 2022, 213, 108681. [CrossRef]
- 94. Ebrahim, M.A.; Aziz, B.A.; Nashed, M.N.F.; Osman, F.A. Optimal design of controllers and harmonic compensators for three-level cascaded control in stationary reference frame for grid-supporting inverters-based AC microgrid. *Energy Rep.* **2022**, *8*, 860–877. [CrossRef]
- 95. Sun, W.; Huang, L.; Liu, Z.; Li, Q.; Zhao, C.; Mu, D. Distributed controller design and stability criterion for microgrids with time-varying delay and rapid switching communication topology. *Sustain. Energy Grids Netw.* **2022**, *29*, 100566. [CrossRef]
- 96. Hossain Lipu, M.S.; Ansari, S.; Miah, M.S.; Hasan, K.; Meraj, S.T.; Faisal, M.; Jamal, T.; Ali, S.H.M.; Hussain, A.; Muttaqi, K.M.; et al. A review of controllers and optimizations based scheduling operation for battery energy storage system towards decarbonization in microgrid: Challenges and future directions. J. Clean. Prod. 2022, 360, 132188. [CrossRef]
- 97. Ziouani, I.; Boukhetala, D.; Darcherif, A.M.; Amghar, B.; Abbassi, I. Hierarchical control for flexible microgrid based on three-phase voltage source inverters operated in parallel. *Int. J. Electr. Power Energy Syst.* **2018**, *95*, 188–201. [CrossRef]
- 98. Bidram, A.; Davoudi, A. Hierarchical structure of microgrids control system. IEEE Trans. Smart Grid 2012, 3, 1963–1976. [CrossRef]
- Rocabert, J.; Luna, A.; Blaabjerg, F.; Rodríguez, P. Control of power converters in AC microgrids. *IEEE Trans. Power Electron.* 2012, 27, 4734–4749. [CrossRef]
- Ilyushin, P.V.; Shepovalova, O.V.; Filippov, S.P.; Nekrasov, A.A. Calculating the Sequence of Stationary Modes in Power Distribution Networks of Russia for Wide-Scale Integration of Renewable Energy Based Installations. *Energy Rep.* 2021, 7, 308–327. [CrossRef]
- Meng, L.; Sanseverino, E.R.; Luna, A.; Dragicevic, T.; Vasquez, J.C.; Guerrero, J.M. Microgrid supervisory controllers and energy management systems: A literature review. *Renew. Sustain. Energy Rev.* 2016, 60, 1263–1273. [CrossRef]
- Yamashita, D.Y.; Vechiu, I.; Gaubert, J.-P. A review of hierarchical control for building microgrids. *Renew. Sustain. Energy Rev.* 2020, 118, 109523. [CrossRef]
- Querol, E.; Romero, F.; Estruch, A.M.; Serrano, J. Design of the Architecture of a Flexible Machining System Using IEC61499 Function Blocks. *Procedia Eng.* 2015, 132, 934–941. [CrossRef]
- 104. Gräßler, I.; Pöhler, A. Implementation of an Adapted Holonic Production Architecture. Procedia CIRP 2017, 63, 138–143. [CrossRef]
- Dawoud, S.M.; Lin, X.; Okba, M.I. Hybrid renewable microgrid optimization techniques: A review. *Renew. Sustain. Energy Rev.* 2018, 82, 2039–2052. [CrossRef]
- 106. Li, J.; Chen, S.; Wu, Y.; Wang, Q.; Liu, X.; Qi, L.; Lu, X.; Gao, L. How to make better use of intermittent and variable energy? A review of wind and photovoltaic power consumption in China. *Renew. Sustain. Energy Rev.* **2021**, *137*, 110626. [CrossRef]
- Zheng, Z.; Yang, S.; Guo, Y.; Jin, X.; Wang, R. Meta-heuristic Techniques in Microgrid Management: A Survey. *Swarm Evol. Comput.* 2023, 78, 101256. [CrossRef]
- 108. Maier, H.R.; Razavi, S.; Kapelan, Z.; Matott, L.S.; Kasprzyk, J.; Tolson, B.A. Introductory overview: Optimization using evolutionary algorithms and other metaheuristics. *Environ. Model. Softw.* **2019**, *114*, 195–213. [CrossRef]
- Papadimitrakis, M.; Giamarelos, N.; Stogiannos, M.; Zois, E.N.; Livanos, N.A.I.; Alexandridis, A. Metaheuristic search in smart grid: A review with emphasis on planning, scheduling and power flow optimization applications. *Renew. Sustain. Energy Rev.* 2021, 145, 111072. [CrossRef]
- 110. Li, W.; Wang, R.; Zhang, T.; Ming, M.; Lei, H. Multi-scenario microgrid optimization using an evolutionary multi-objective algorithm. *Swarm Evol. Comput.* **2019**, *50*, 100570. [CrossRef]
- Bharothu, J.N.; Sridhar, M.; Rao, R.S. Modified adaptive differential evolution based optimal operation and security of AC-DC microgrid systems. *Int. J. Electr. Power Energy Syst.* 2018, 103, 185–202. [CrossRef]
- 112. Tavana, M.; Kazemi, M.R.; Vafadarnikjoo, A.; Mobin, M. An Artificial Immune Algorithm for Ergonomic Product Classification Using Anthropometric Measurements. *Measurement* **2016**, *93*, 621–629. [CrossRef]
- 113. Katsigiannis, Y.A.; Georgilakis, P.S.; Karapidakis, E.S. Hybrid simulated annealing-tabu search method for optimal sizing of autonomous power systems with renewables. *IEEE Trans. Sustain. Energy* **2012**, *3*, 330–338. [CrossRef]
- 114. He, J.; Li, Y.W.; Guerrero, J.M.; Blaabjerg, F.; Vasquez, J.C. An islanding Microgrid power sharing approach using enhanced virtual impedance control scheme. *IEEE Trans. Power Electron.* **2013**, *28*, 5272–5282. [CrossRef]
- 115. Suvorov, A.; Askarov, A.; Bay, Y.; Maliuta, B.; Achitaev, A.; Suslov, K. Comparative small-signal stability analysis of voltagecontrolled and enhanced current-controlled virtual synchronous generators under weak and stiff grid conditions. *Int. J. Electr. Power Energy Syst.* **2023**, *147*, 108891. [CrossRef]
- Merritt, N.; Chakraborty, C.; Bajpai, P. New voltage control strategies for VSC-based DG units in an unbalanced microgrid. *IEEE Trans. Sustain. Energy* 2017, 8, 1127–1139. [CrossRef]
- Gong, H.; Wang, X.; Harnefors, L. Rethinking Current Controller Design for PLL-Synchronized VSCs in Weak Grids. *IEEE Trans.* Power Electron. 2022, 37, 1369–1381. [CrossRef]

- Wang, X.; Taul, M.G.; Wu, H.; Liao, Y.; Blaabjerg, F.; Harnefors, L. Grid-Synchronization Stability of Converter-Based Resources— An Overview. *IEEE Open J. Ind. Appl.* 2020, 1, 115–134. [CrossRef]
- Diaz-Sanahuja, C.; Peñarrocha-Alós, I.; Vidal-Albalate, R. Multivariable phase-locked loop free strategy for power control of grid-connected voltage source converters. *Electr. Power Syst. Res.* 2022, 210, 108084. [CrossRef]
- Lasseter, R.H.; Chen, Z.; Pattabiraman, D. Grid-Forming Inverters: A Critical Asset for the Power Grid. *IEEE J. Emerg. Sel. Top.* Power Electron. 2020, 8, 925–935. [CrossRef]
- 121. Pan, D.; Wang, X.; Liu, F.; Shi, R. Transient Stability of Voltage-Source Converters with Grid-Forming Control: A Design-Oriented Study. *IEEE J. Emerg. Sel. Top. Power Electron.* 2020, *8*, 1019–1033. [CrossRef]
- 122. Harnefors, L.; Bongiorno, M.; Lundberg, S. Input-admittance calculation and shaping for controlled voltage-source converters. *IEEE Trans. Ind. Electron.* 2007, *54*, 3323–3334. [CrossRef]
- 123. Zou, C.; Rao, H.; Xu, S.; Li, Y.; Li, W.; Chen, J.; Zhao, X.; Yang, Y.; Lei, B. Analysis of Resonance between a VSC-HVDC Converter and the AC Grid. *IEEE Trans. Power Electron.* **2018**, *33*, 10157–10168. [CrossRef]
- 124. Ayari, M.; Belhaouane, M.; Guillaud, X.; Braiek, N.B. Multivariable Optimal PID control design for Modular Multilevel Converter. In Proceedings of the Conférence Internationale en Sciences et Technologies Electriques, Marrakech, Maroc, 25–28 October 2016; Available online: https://lilloa.univ-lille.fr/handle/20.500.12210/31450 (accessed on 9 January 2023).
- 125. Li, J.; Konstantinou, G.; Wickramasinghe, H.R.; Pou, J. Operation and Control Methods of Modular Multilevel Converters in Unbalanced AC Grids: A Review. *IEEE J. Emerg. Sel. Top. Power Electron.* 2019, 7, 1258–1271. [CrossRef]
- 126. Wang, J.; Bo, D.; Miao, Q.; Li, Z.; Wu, X.; Lv, D. Maximum power point tracking control for a doubly fed induction generator wind energy conversion system based on multivariable adaptive super-twisting approach. *Int. J. Electr. Power Energy Syst.* 2021, 124, 106347. [CrossRef]
- 127. Zhang, B.; Fu, X. Improved droop control strategy based on voltage feedforward current compensation. *Energy Rep.* 2021, 7, 434–441. [CrossRef]
- 128. Khan, M.Z.; Ahmed, E.M.; Habib, S.; Ali, Z.M. Multi-objective Optimization Technique for Droop Controlled Distributed Generators in AC Islanded Microgrid. *Electr. Power Syst. Res.* **2022**, *213*, 108671. [CrossRef]
- 129. Sinha, S.; Ghosh, S.; Bajpai, P. Power sharing through interlinking converters in adaptive droop controlled multiple microgrid system. *Int. J. Electr. Power Energy Syst.* 2021, 128, 106649. [CrossRef]
- Wang, J.; Zhang, X.; Zhu, Q.; Chen, S.; Peng, B. Transient stability enhancement control strategy for droop-controlled voltage source converter. *Energy Rep.* 2022, *8*, 35–44. [CrossRef]
- Tayab, U.B.; Roslan, M.A.; Hwai, L.J.; Kashif, M. A review of droop control techniques for microgrid. *Renew. Sustain. Energy Rev.* 2017, 76, 717–727. [CrossRef]
- 132. Sheikhzadehbaboli, P.; Samimi, A.; Ebadi, M.; Bayat, M.; Pirayesh, A. Frequency control in standalone renewable based-microgrids using steady state load shedding considering droop characteristic. *Int. J. Electr. Power Energy Syst.* 2022, 142, 108351. [CrossRef]
- 133. Sun, P.; Wang, Y.; Khalid, M.; Blasco-Gimenez, R.; Konstantinou, G. Steady-state power distribution in VSC-based MTDC systems and dc grids under mixed P/V and I/V droop control. *Electr. Power Syst. Res.* **2023**, *214*, 108798. [CrossRef]
- 134. Wang, B.; Sang, Y. Dual-mode operation control of smart micro grid based on droop strategy. *Energy Rep.* **2022**, *8*, 9017–9024. [CrossRef]
- 135. Datta, U.; Shi, J.; Kalam, A. Primary frequency control of a microgrid with integrated dynamic sectional droop and fuzzy based pitch angle control. *Int. J. Electr. Power Energy Syst.* **2019**, *111*, 248–259. [CrossRef]
- 136. Li, Y.; Li, Y.W. Power management of inverter interfaced autonomous microgrid based on virtual frequency-voltage frame. *IEEE Trans. Smart Grid* **2011**, 2, 30–40. [CrossRef]
- 137. Dokus, M.; Stallmann, F.; Mertens, A. Sequence Impedance-Based Stability Analysis of AC Microgrids Controlled by Virtual Synchronous Generator Control Methods. *IFAC-Pap. OnLine* **2020**, *53*, 12221–12228. [CrossRef]
- 138. Zhang, X.; Zhang, Y.; Fang, R.; Xu, D. An improved virtual inductance control method considering PLL dynamic based on impedance modeling of DFIG under weak grid. *Int. J. Electr. Power Energy Syst.* **2020**, *118*, 105772. [CrossRef]
- 139. Lin, Y.; Fu, L. A novel virtual admittance droop based inertial coordination control for medium-voltage direct current ship with hybrid energy storage. *J. Energy Storage* **2022**, *56*, 105962. [CrossRef]
- Rezaei, N.; Kalantar, M. Smart microgrid hierarchical frequency control ancillary service provision based on virtual inertia concept: An integrated demand response and droop controlled distributed generation framework. *Energy Convers. Manag.* 2015, 92, 287–301. [CrossRef]
- Vyawahare, D. Dynamics of power flow in a stand-alone microgrid using four-leg inverters for nonlinear and unbalanced loads. In *Distributed Energy Resources in Microgrids: Integration, Challenges and Optimization;* Academic Press: Cambridge, MA, USA, 2019; pp. 113–141. [CrossRef]
- 142. Naderipour, A.; Abdul-Malek, Z.; Ramachandaramurthy, V.K.; Kalam, A.; Miveh, M.R. Hierarchical control strategy for a three-phase 4-wire microgrid under unbalanced and nonlinear load conditions. *ISA Trans.* **2019**, *94*, 352–369. [CrossRef] [PubMed]
- 143. Pinthurat, W.; Hredzak, B.; Konstantinou, G.; Fletcher, J. Techniques for compensation of unbalanced conditions in LV distribution networks with integrated renewable generation: An overview. *Electr. Power Syst. Res.* 2023, 214, 108932. [CrossRef]
- 144. Jiang, X.S.; Hong, Y.; Chen, K.; Cheung, N.; Wai Eric Cheng, K.; Pan, J.F. A communication-free droop control scheme for distributed linear switched reluctance wave power generation system with adaptive voltage compensation. *Int. J. Electr. Power Energy Syst.* 2022, 143, 108461. [CrossRef]

- 145. Yang, D.; Jin, Z.; Zheng, T.; Jin, E. An adaptive droop control strategy with smooth rotor speed recovery capability for type III wind turbine generators. Int. J. Electr. Power Energy Syst. 2022, 135, 107532. [CrossRef]
- 146. Ouammi, A.; Zejli, D. Centralized controller for the optimal operation of a cooperative cluster of connected microgrids powered multi-greenhouses. *Sustain. Comput. Inform. Syst.* 2022, 33, 100641. [CrossRef]
- Shanmugasundaram, S.; Kashkynbayev, A.; Udhayakumar, K.; Rakkiyappan, R. Centralized and decentralized controller design for synchronization of coupled delayed inertial neural networks via reduced and non-reduced orders. *Neurocomputing* 2022, 469, 91–104. [CrossRef]
- 148. Raju, P.; Jain, T. Robust optimal centralized controller to mitigate the small signal instability in an islanded inverter based microgrid with active and passive loads. *Int. J. Electr. Power Energy Syst.* **2017**, *90*, 225–236. [CrossRef]
- 149. Chen, Q.; Luan, X.; Liu, F. Analytical Design of Centralized PI Controller for High Dimensional Multivariable Systems. *IFAC Proc. Vol.* **2013**, *46*, 643–648. [CrossRef]
- 150. Yang, Y.; Feng, X.; Li, J.; Hua, C. Robust fixed-time cooperative control strategy design for nonlinear multiple-master/multipleslave teleoperation system. *J. Frankl. Inst.* **2022**, *360*, 2193–2214. [CrossRef]
- Yang, J.; Yuan, W.; Sun, Y.; Han, H.; Hou, X.; Guerrero, J.M. A novel quasi-master-slave control frame for PV-storage independent microgrid. *Int. J. Electr. Power Energy Syst.* 2018, 97, 262–274. [CrossRef]
- 152. Pena Ramirez, J.; Garcia, E.; Alvarez, J. Master-slave synchronization via dynamic control. *Commun. Nonlinear Sci. Numer. Simul.* 2020, *80*, 104977. [CrossRef]
- 153. Shushpanov, I.; Suslov, K.; Ilyushin, P.; Sidorov, D. Towards the flexible distribution networks design using the reliability performance metric. *Energies* **2021**, *14*, 6193. [CrossRef]
- 154. Ruban, N.; Suvorov, A.; Andreev, M.; Ufa, R.; Askarov, A.; Gusev, A.; Bhalja, B. Software and Hardware Decision Support System for Operators of Electrical Power Systems. *IEEE Trans. Power Syst.* 2021, *36*, 3840–3848. [CrossRef]
- Ilyushin, P.V.; Pazderin, A.V. Approaches to organization of emergency control at isolated operation of energy areas with distributed generation. In Proceedings of the International Ural Conference on Green Energy, Chelyabinsk, Russia, 4–6 October 2018. [CrossRef]
- Al-Shetwi, A.Q.; Hannan, M.A.; Jern, K.P.; Mansur, M.; Mahlia, T.M.I. Grid-connected renewable energy sources: Review of the recent integration requirements and control methods. J. Clean. Prod. 2020, 253, 119831. [CrossRef]
- 157. Meral, M.E.; Çelík, D. A comprehensive survey on control strategies of distributed generation power systems under normal and abnormal conditions. *Annu. Rev. Control* 2019, 47, 112–132. [CrossRef]
- 158. Bidram, A.; Davoudi, A.; Lewis, F.L.; Guerrero, J.M. Distributed cooperative secondary control of microgrids using feedback linearization. *IEEE Trans. Power Syst.* 2013, 28, 3462–3470. [CrossRef]
- 159. Hou, S.; Chen, J.; Chen, G. Distributed control strategy for voltage and frequency restoration and accurate reactive power-sharing for islanded microgrid. *Energy Rep.* **2023**, *9*, 742–751. [CrossRef]
- Samadi, E.; Badri, A.; Ebrahimpour, R. Decentralized multi-agent based energy management of microgrid using reinforcement learning. Int. J. Electr. Power Energy Syst. 2020, 122, 106211. [CrossRef]
- Raya-Armenta, J.M.; Bazmohammadi, N.; Avina-Cervantes, J.G.; Sáez, D.; Vasquez, J.C.; Guerrero, J.M. Energy management system optimization in islanded microgrids: An overview and future trends. *Renew. Sustain. Energy Rev.* 2021, 149, 111327. [CrossRef]
- Sharma, P.; Dutt Mathur, H.; Mishra, P.; Bansal, R.C. A critical and comparative review of energy management strategies for microgrids. *Appl. Energy* 2022, 327, 120028. [CrossRef]
- 163. Zhan, S.; Chong, A. Data requirements and performance evaluation of model predictive control in buildings: A modeling perspective. *Renew. Sustain. Energy Rev.* 2021, 142, 110835. [CrossRef]
- 164. Yao, Y.; Shekhar, D.K. State of the art review on model predictive control (MPC) in Heating Ventilation and Air-conditioning (HVAC) field. *Build. Environ.* 2021, 200, 107952. [CrossRef]
- Mariano-Hernández, D.; Hernández-Callejo, L.; Zorita-Lamadrid, A.; Duque-Pérez, O.; Santos García, F. A review of strategies for building energy management system: Model predictive control, demand side management, optimization, and fault detect & diagnosis. J. Build. Eng. 2021, 33, 101692. [CrossRef]
- 166. Hirsch, A.; Parag, Y.; Guerrero, J. Microgrids: A review of technologies, key drivers, and outstanding issues. *Renew. Sustain. Energy Rev.* 2018, 90, 402–411. [CrossRef]
- 167. Shi, Q.; Li, F.; Hu, Q.; Wang, Z. Dynamic demand control for system frequency regulation: Concept review, algorithm comparison, and future vision. *Electr. Power Syst. Res.* **2018**, 154, 75–87. [CrossRef]
- 168. Akagi, S.; Yoshizawa, S.; Ito, M.; Fujimoto, Y.; Miyazaki, T.; Hayashi, Y.; Tawa, K.; Hisada, T.; Yano, T. Multipurpose control and planning method for battery energy storage systems in distribution network with photovoltaic plant. *Int. J. Electr. Power Energy Syst.* 2020, 116, 105485. [CrossRef]
- Taghieh, A.; Mohammadzadeh, A.; Zhang, C.; Kausar, N.; Castillo, O. A type-3 fuzzy control for current sharing and voltage balancing in microgrids. *Appl. Soft Comput.* 2022, 129, 109636. [CrossRef]
- 170. Karimi, H.; Beheshti, M.T.H.; Ramezani, A.; Zareipour, H. Intelligent control of islanded AC microgrids based on adaptive neuro-fuzzy inference system. *Int. J. Electr. Power Energy Syst.* 2021, 133, 107161. [CrossRef]
- Radich, A.S.; Magalhães Filho, S.C.; Borges, R.F.O.; Scheid, C.M.; Meleiro, L.A.C.; Calçada, L.A. Real-time control and monitoring of drilling fluids density by fuzzy-based control system. *Geoenergy Sci. Eng.* 2023, 222, 211421. [CrossRef]

- 172. Hu, W.; Sun, Z.; Zhang, Y.; Li, Y. Joint Manufacturing and Onsite Microgrid System Control Using Markov Decision Process and Neural Network Integrated Reinforcement Learning. *Procedia Manuf.* **2019**, *39*, 1242–1249. [CrossRef]
- 173. Kaushal, J.; Basak, P. Power quality control based on voltage sag/swell, unbalancing, frequency, THD and power factor using artificial neural network in PV integrated AC microgrid. *Sustain. Energy Grids Netw.* **2020**, *23*, 100365. [CrossRef]
- 174. Katiraei, F.; Iravani, R.; Hatziargyriou, N.; Dimeas, A. Microgrid management control and operation aspects of microgrids. *IEEE Power Energy Mag.* 2008, *6*, 54–65. [CrossRef]
- 175. Ilyushin, P.V.; Kulikov, A.L.; Filippov, S.P. Adaptive algorithm for automated undervoltage protection of industrial power districts with distributed generation facilities. In Proceedings of the 2019 International Russian Automation Conference (RusAutoCon), Sochi, Russia, 8–14 September 2019. [CrossRef]
- 176. Hojabri, M.; Toudeshki, A.; Hojabri, M.; Toudeshki, A. Power Quality Consideration for Off-Grid Renewable Energy Systems. *Energy Power Eng.* **2013**, *5*, 377–383. [CrossRef]
- 177. Rajan, R.; Fernandez, F.M.; Yang, Y. Primary frequency control techniques for large-scale PV-integrated power systems: A review. *Renew. Sustain. Energy Rev.* **2021**, 144, 110998. [CrossRef]
- Wang, J.; Jin, C.; Wang, P. A Uniform Control Strategy for the Interlinking Converter in Hierarchical Controlled Hybrid AC/DC Microgrids. *IEEE Trans. Ind. Electron.* 2018, 65, 6188–6197. [CrossRef]
- 179. Ma, D.; Sun, Q.; Xie, X.; Li, X. Event triggering power sharing control for AC/DC microgrids based on P -F droop curve method. *J. Frankl. Inst.* **2019**, 356, 1225–1246. [CrossRef]
- 180. Chanfreut, P.; Maestre, J.M.; Camacho, E.F. A survey on clustering methods for distributed and networked control systems. *Annu. Rev. Control* **2021**, *52*, 75–90. [CrossRef]
- Bandeiras, F.; Pinheiro, E.; Gomes, M.; Coelho, P.; Fernandes, J. Review of the cooperation and operation of microgrid clusters. *Renew. Sustain. Energy Rev.* 2020, 133, 110311. [CrossRef]
- Ilyushin, P.V.; Filippov, S.P. Under-frequency load shedding strategies for power districts with distributed generation. In Proceedings of the 2019 International Conference on Industrial Engineering, Applications and Manufacturing (ICIEAM), Sochi, Russia, 25–29 March 2019. [CrossRef]
- 183. Kim, I.; Harley, R.G. The transient-state effect of the reactive power control of photovoltaic systems on a distribution network. *Int. J. Electr. Power Energy Syst.* 2018, 99, 630–637. [CrossRef]
- 184. Ratnam, K.S.; Palanisamy, K.; Yang, G. Future low-inertia power systems: Requirements, issues, and solutions—A review. *Renew. Sustain. Energy Rev.* 2020, 124, 109773. [CrossRef]
- 185. Wang, Y.; Wang, L.; Li, M.; Chen, Z. A review of key issues for control and management in battery and ultra-capacitor hybrid energy storage systems. *ETransportation* **2020**, *4*, 100064. [CrossRef]
- Ortiz, L.; Gutiérrez, L.B.; González, J.W.; Águila, A. A novel strategy for dynamic identification in AC/DC microgrids based on ARX and Petri Nets. *Heliyon* 2020, 6, e03559. [CrossRef] [PubMed]
- Kyriakarakos, G.; Dounis, A.I.; Arvanitis, K.G.; Papadakis, G. A fuzzy cognitive maps-petri nets energy management system for autonomous polygeneration microgrids. *Appl. Soft Comput.* 2012, 12, 3785–3797. [CrossRef]
- Zia, M.F.; Elbouchikhi, E.; Benbouzid, M. Microgrids energy management systems: A critical review on methods, solutions, and prospects. *Appl. Energy* 2018, 222, 1033–1055. [CrossRef]
- 189. Ilyushin, P.V. Analysis of the specifics of selecting relay protection and automatic (RPA) equipment in distributed networks with auxiliary low-power generating facilities. *Power Technol. Eng.* **2018**, *51*, 713–718. [CrossRef]
- Li, Q.; Gao, M.; Lin, H.; Chen, Z.; Chen, M. MAS-based distributed control method for multi-microgrids with high-penetration renewable energy. *Energy* 2019, 171, 284–295. [CrossRef]
- 191. Muyeen, S.M.; Islam, S.M.; Blaabjerg, F. Variability, scalability and stability of microgrids. *Var. Scalability Stab. Microgrids* **2019**, 1–623. [CrossRef]
- 192. Harmouch, F.Z.; Krami, N.; Hmina, N. A multiagent based decentralized energy management system for power exchange minimization in microgrid cluster. *Sustain. Cities Soc.* 2018, 40, 416–427. [CrossRef]
- Blanke, M.; Kinnaert, M.; Lunze, J.; Staroswiecki, M. Diagnosis and Fault-Tolerant Control. In *Diagnosis and Fault-Tolerant Control*; Springer: Berlin/Heidelberg, Germany, 2003. [CrossRef]
- 194. Rahiminejad, A.; Ghafouri, M.; Atallah, R.; Lucia, W.; Debbabi, M.; Mohammadi, A. Resilience enhancement of Islanded Microgrid by diversification, reconfiguration, and DER placement/sizing. *Int. J. Electr. Power Energy Syst.* **2023**, 147, 108817. [CrossRef]
- 195. Kulikov, A.L.; Ilyushin, P.V.; Suslov, K.V.; Karamov, D.N. Coherence of digital processing of current and voltage signals at decimation for power systems with a large share of renewable power stations. *Energy Reports* **2022**, *8*, 1464–1478. [CrossRef]
- Bozalakov, D.V.; Laveyne, J.; Desmet, J.; Vandevelde, L. Overvoltage and voltage unbalance mitigation in areas with high penetration of renewable energy resources by using the modified three-phase damping control strategy. *Electr. Power Syst. Res.* 2019, *168*, 283–294. [CrossRef]
- 197. Hernández-Callejo, L.; Gallardo-Saavedra, S.; Alonso-Gómez, V. A review of photovoltaic systems: Design, operation and maintenance. *Sol. Energy* **2019**, *188*, 426–440. [CrossRef]
- 198. Jithin, S.; Rajeev, T. Investigation on Microgrid Control and Stability. Smart Grids Microgrids 2022, 99–125. [CrossRef]
- 199. Ajewole, T.O.; Olabode, O.E.; Babalola, O.S.; Omoigui, M.O. Use of experimental test systems in the application of electric microgrid technology across the sub-Saharan Africa: A review. *Sci. Afr.* **2020**, *8*, e00435. [CrossRef]

- 200. Jiayi, H.; Chuanwen, J.; Rong, X. A review on distributed energy resources and MicroGrid. *Renew. Sustain. Energy Rev.* 2008, 12, 2472–2483. [CrossRef]
- Case Studies of Smart Community Demonstration Project | NEDO. (n.d.). Available online: https://www.nedo.go.jp/english/ news/reports_20130222.html#India (accessed on 14 March 2023).
- Krishnamurthy, S.; Jahns, T.M.; Lasseter, R.H. The operation of diesel gensets in a CERTS microgrid. In Proceedings of the IEEE Power and Energy Society 2008 General Meeting: Conversion and Delivery of Electrical Energy in the 21st Century, PES, Pittsburgh, PA, USA, 20–24 July 2008. [CrossRef]
- Bossi, C.; Degner, T.; Tselepis, S. Distributed Generation with High Penetration of Renewable Energy Sources. 2006. Available online: https://scholar.google.com/citations?view_op=view_citation&hl=en&user=wwOk4OoAAAAJ&citation_for_view= wwOk4OoAAAAJ:M3ejUd6NZC8C (accessed on 9 January 2023).
- Jayawarna, N.; Jones, C.; Barnes, M.; Jenkins, N. Operating MicroGrid energy storage control during network faults. In Proceedings of the 2007 IEEE International Conference on System of Systems Engineering, SOSE, San Antonio, TX, USA, 16–18 April 2007. [CrossRef]
- Shamarova, N.; Suslov, K.; Ilyushin, P.; Shushpanov, I. Review of Battery Energy Storage Systems Modeling in Microgrids with Renewables Considering Battery Degradation. *Energies* 2022, 15, 6967. [CrossRef]
- Martin-Martínez, F.; Sánchez-Miralles, A.; Rivier, M. A literature review of Microgrids: A functional layer based classification. *Renew. Sustain. Energy Rev.* 2016, 62, 1133–1153. [CrossRef]
- 207. Zhao, Y.; Song, X.; Wang, F.; Cui, D. Multiobjective optimal dispatch of microgrid based on analytic hierarchy process and quantum particle swarm optimization. *Glob. Energy Interconnect.* **2020**, *3*, 562–570. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.